

UMass Global UMass Global ScholarWorks

Dissertations

Winter 11-28-2022

Designing and Delivering K-12 Education Professional Development: A Delphi Study

Christalle Hart chart3@mail.umassglobal.edu

Follow this and additional works at: https://digitalcommons.umassglobal.edu/edd_dissertations

Part of the Educational Leadership Commons, and the Teacher Education and Professional Development Commons

Recommended Citation

Hart, Christalle, "Designing and Delivering K-12 Education Professional Development: A Delphi Study" (2022). *Dissertations*. 481.

https://digitalcommons.umassglobal.edu/edd_dissertations/481

This Dissertation is brought to you for free and open access by UMass Global ScholarWorks. It has been accepted for inclusion in Dissertations by an authorized administrator of UMass Global ScholarWorks. For more information, please contact christine.bombaro@umassglobal.edu.

Designing and Delivering K-12 Education Professional Development: A Delphi Study

A Dissertation by

Christalle A. Hart

University of Massachusetts Global

Irvine, California

School of Education

Submitted in partial fulfillment of the requirements for the degree of

Doctor of Education in Organizational Leadership

November 2022

Committee in charge:

Philip Pendley, Ed.D., Committee Chair

Steve Kong, Ed.D.

Martinrex Kedziora, Ed.D.

University of Massachusetts Global

A Nonprofit Affiliate of the University of Massachusetts

Doctor of Education in Organizational Leadership

The dissertation of Christalle A. Hart is approved.

Sec	, Dissertation Chair
Philip Pendley, Ed.D.	
Jets	, Committee Member
Steve Kong, Ed.D.	
Martinrex Kedziora, Ed.D.	, Committee Member
Peters amount	, Associate Dean
Patrick Ainsworth, Ed.D.	

November 2022

Designing and Delivering K-12 Education Professional Development: A Delphi Study

Copyright © 2022

by Christalle A. Hart

ACKNOWLEDGEMENTS

I would like to express my immense gratitude to my faculty chair advisor, Dr. Pendley. Thank you for sharing your knowledge, expertise, guidance, and continual encouragement throughout this doctoral journey. I looked forward to each exchange, in which you closed with, "One step closer." All of the steps added up! I also want to thank my committee member, Dr. Kong who provided constant support, meaningful feedback and patient counsel throughout this doctoral journey. Dr. Kedziora, thank you for being a wonderful cohort mentor and committee member. Your unending positivity and ability to connect experiences to learning through storytelling perpetually uplifted me. I am grateful for this leadership team. You all have shaped me, inspired me, and I have learned so much along the way.

I also want to thank my family for their love, support, and grace as I progressed through this journey. Much credit is given to my husband Russell, who recognized my dream and provided the space and time for me to fulfill it. I thank my three children, Andy, Miranda, and Payton who brought pockets of joy in between the hours of research. I am looking forward to our many adventures in the future now that this journey is accomplished.

Thank you to the communities who have championed me through the completion of this dissertation. The Riverside IOTAs, my workout Squad, the Best of Friends ladies, and my daily accountability partner, Caryn, and my ILE team, who all gave me support and continual encouragement throughout this journey.

Now, I look forward to being "one step closer" to coffee on the front porch and pondering the next journey to embark on.

ABSTRACT

Designing and Delivering K-12 Education Professional Development: A Delphi Study by Christalle A. Hart

Purpose: This futures Delphi study aims to determine what experts in professional development predict the delivery and content of K-12 education and how K-12 educational organizations will need to structure themselves to deliver predicted professional development in 2026 and beyond. Also, the purpose was to have the experts rate the likelihood of the predictions made to be enacted by 2026 and beyond.

Methodology: The research design for this study used a Delphi approach which allowed for data collection from a panel of experts in K-12 education leading professional development in a public school district. The Delphi study collected data during three rounds of electronic surveys. The electronic surveys contained open-ended questions in Round 1 and 3, while Round 2 included a Likert scale rating.

Findings: The initial round of data collection returned 26 predictions for professional development delivery and content, which then needed to be rated. Round 2 rated the likelihood of implementation for the top five delivery methods and top five content areas. **Conclusion:** The study had three major conclusions. The first was associated with the importance of providing differentiated professional development to all learners. The second conclusion indicated that professional development should be data driven. The final conclusion highlighted the importance of embedding instructional best practices in professional development.

Recommendation for Action: Nine total recommendations for future research are provided to advance the body of literature and knowledge around designing and delivering professional development in K-12 education districts.

TABLE OF CONTENTS

CHAPTER I: INTRODUCTION	1
Background	3
History of Professional Development and Education Policy	3
California Teacher Standards	
Theoretical Frameworks	6
Adult Learning Theory	7
Professional Development Models	
The Changing Needs of the Teacher	
Statement of Research Problem	
Purpose Statement	11
Research Questions	
Significance of the Study	
Definitions	
Delimitations	
Organization of the Study	
CHAPTER II: REVIEW OF THE LITERATURE	16
Literature Review	16
The Role of Professional Development	17
Historical Perspective of K-12 Professional Development	19
History of Professional Development: 1800s to 1960s	
History of Professional Development: 1970s to Present	20
The Impact of Federal Reforms on Professional Development Policies	21
California Standards for the Teaching Profession	
Professional Development Effective Practices	25
Professional Development Delivery Models	27
Workshops and Conferences	
District Provided Learning Opportunities	28
Coaching Model	
Professional Learning Community	
Professional Learning Networks	
Barriers to Effective Professional Development	34
Challenges for Determining the Effectiveness of Professional Development	36
Adult Learning Theory	
21st Century Learning	38
Research Gap	40
Summary	41
CHAPTER III: METHODOLOGY	44
Overview	
Purpose Statement	
Research Questions	
Research Design	
Population	
Sampling Frame	47

Sample	47
Sample Selection Process	48
Instrumentation	49
Round 1	49
Round 2	49
Round 3	
Validity	50
Reliability	
Data Collection	
Round 1	
Round 2	
Round 3	
Data Analysis	
Round 1	
Round 2	
Round 3	
Limitations	
Summary	
CHAPTER IV: RESEARCH, DATA COLLECTION, AND FINDINGS	56
Overview	56
Purpose Statement	56
Research Questions	57
Research Methods and Data Collection Procedures	
Population and Sample	59
Presentation of the Data	60
Research Question 1	60
Round 1	60
Analysis of Round 1	61
Research Question 2	64
Round 2	65
Analysis of round 2	66
Research Question 3	70
Round 3	70
Analysis of Round 3	71
Research Question 4	82
Research Question 4 Analysis	82
Summary	85
CHAPTER V: FINDINGS, CONCLUSIONS, AND RECOMMENDATIONS	QC
Purpose Statement	
Research Questions	
Methodology	
Major Findings	
Research Questions	
Research Question 1	
Round 1	90

Research Question 2	92
Round 2	
Research Question 3	92
Research Question 4	93
Round 3	93
Prediction 1: Professional Development Delivery Models will Deliver	
Content Through a Blended (synchronous/asynchronous) Model	93
Prediction 2: Professional Development Delivery Models will be Data-	
Driven	94
Prediction 3: Professional Development Delivery Models will Focus on Be	st
Practices for Instruction	
Prediction 4: Professional Development Delivery Models will Provide Self	
Paced, Asynchronous Learning	
Prediction 5: Professional Development Delivery Models will Deliver	
Professional Development with Integrated Content	95
Prediction 6: Professional Development Delivery Models will Have Conter	
That is Driven by District Initiatives and Priorities	96
Prediction 7: Professional Development Delivery Models that will Determi	ne
Content After Engaging in Data Analysis	
Prediction 8: Professional Development Content will Emphasize Skills	98
Prediction 9: Professional Development Content will be Differentiated for	all
Learners	98
Prediction 10: Professional Development Content That is Universally	
Designed For All Learners	99
Unexpected Findings	. 101
Conclusions	. 102
Conclusion 1	. 103
Conclusion 2	. 103
Conclusion 3	. 103
Implications for Action	. 104
Implications for Action 1	. 104
Implications for Action 2	. 105
Implications for Action 3	. 105
Recommendations for Further Research	. 106
Concluding Remarks and Reflections	. 107
REFERENCES	. 108
APPENDICES	. 120

LIST OF TABLES

Table 1.	List of Possible Professional Development Attributes for Delivery Models 62
Table 2.	List of Possible Professional Development Content Areas
Table 3.	Mean Ranking of Professional Development Attributes for Delivery Models 65
Table 4.	Mean Ranking of Professional Development Content Areas
Table 5.	Highest Rated Professional Development Delivery 67
Table 6.	Highest Rated Professional Development Content Prediction
Table 7.	Lowest Ranked Professional Development Delivery and Content 69
Table 8.	Top Five K-12 Professional Development Predictions for Professional
	Development Delivery and Content
Table 9.	Top Structures for Implementing K-12 Professional Development Predictions
	for Blended Professional Development Delivery
Table 10.	Top Structures for Implementing K-12 Professional Development Predictions
	for Data-driven Professional Development Delivery
Table 11.	Top Structures for Implementing K-12 Professional Development Predictions
	for Professional Development that focuses on Best Practices for Instruction . 75
Table 12.	Top Structures for Implementing K-12 Professional Development Predictions
	for Self-paced, Asynchronous Learning
Table 13.	Top Structures for Implementing K-12 Professional Development Predictions
	for a Model Focused on Integrated Content
Table 14.	Top Structures for Implementing Professional Development Content Driven by
	District Initiatives and Priorities

Table 15. Top Structures for Implementing K-12 Profession	onal Development Content
Driven by Data Analysis	79
Table 16. Top Structures for Implementing K-12 Profession	onal Development Predictions
for Professional Development content with an E	Emphasis on Skill 80
Table 17. Top Structures for Implementing Differentiated	Content in K-12 Professional
Development	
Table 18. Top Structures for Implementing K-12 Profession	onal Development Content that
is Universally Designed	82
Table 19. Expert Responses to Predictions	85

CHAPTER I: INTRODUCTION

Education aims to increase student learning and prepare students to contribute to society (Vadeboncoeur, 2005). Educators must prepare students for a future we may not imagine, solving problems that have not been discovered, and using technologies that do not exist. The Future of Jobs Survey (World Economic Forum, 2020) listed the top skills for 2025 employers sought: (a) analytical thinking, (b) innovation, (c) complex problem solving, (d) critical thinking, (e) creativity, (f) emotional intelligence, and (g) resilience. To better prepare students to be contributors to society, educators must provide ample opportunities to apply 21st century learning skills such as (a) communication, (b) collaboration, (c) critical thinking, and (d) creativity in their daily instructional experiences. Educational reform spurred many opportunities for increased teacher professional development (PD) through additional funding.

As educational reform progressed, the standards for student learning became increasingly rigorous. In 1965 the Elementary and Secondary Education Act (ESEA) was established by President Lyndon Johnson, a former school teacher. Title II allocates most funds to state and district-level activities to improve teacher and student learning (McKnight, 2018). A Nation at Risk (U.S. Department of Education [DoE], 2008) recommended new rigorous standards that required students to meet higher-level requirements to graduate. Recent educational reforms, including No Child Left Behind (NCLB) (DoE, 2007, 2017), and the Every Student Succeeds Act (DoE, n.d.), have addressed the importance of continued educator development. Teachers have the highest impact on student learning; therefore, it is critical to provide high-quality professional

learning opportunities to increase their quality and effectiveness (Hattie, 2012; Reyna, 2019).

The California Standards for the Teaching Profession (CSTPs) were developed in the 1990s and intended to provide new educators with a framework for formative assessment and best practices. The standards are organized into six interrelated domains of teaching practice. These standards include:

- Supporting and engaging learners.
- Effective learning environments.
- Subject matter knowledge.
- Designing learning experiences.
- Assessment.
- Developing as a professional educator.

These six areas allow educators to reflect on their practice and student learning, create professional goals to increase student learning, and assess their progress towards these goals (Commission on Teacher Credentialing, 2009). The CSTPs acknowledge that teachers are never finished with their professional learning because they teach a diverse student population in a rapidly changing world (Commission on Teacher Credentialing, 2009). However, there are discrepancies between the teaching and learning educators provide their students and the teaching they experience with their professional learning (Hunzicker, 2011; Tate, 2012).

Traditionally, PD consists of fragmented learning, one-day sit-and-get workshops, and in-service meetings that do not connect to a school-wide or individualized plan (Dede, 2006; Tate, 2012). Adult learners need professional learning to be relevant and

purposeful. The learning should also build upon life experiences (Knowles et al., 2005). PD should be ongoing, intensive, job-embedded, data-driven, collaborative, personalized, and classroom-focused (Tate, 2012; Tucker, 2006). In addition, school districts need a professional growth and improvement system, which is essential for increasing student achievement.

Background

Today's diverse student population is rapidly evolving, and teachers need ongoing support to foster excellence in teaching and learning. PD is a critical component to increased student achievement (Yoon et al., 2007). Hattie (2012) argued that teachers are the most important factor influencing student achievement. PD affected student achievement through enhanced teacher knowledge, skills, and improved teaching practices (Bastin, 2003; Guskey & Yoon, 2009; Yoon et al., 2007). Theories of adult learning enrich the design and delivery of PD, which is also a critical component to the success of the PD. PD is continually evolving; however, there is little research on identifying and outlining the most effective PD activities utilized in K-12 districts that have allowed PD administrators to reach expert status and subsequently impact student achievement. It is essential to understand the impact of effective PD systems in K-12 districts and bridge the gap between theory and practice in organizations.

History of Professional Development and Education Policy

The methods for planning and providing PD have dramatically changed over the years. In the 1800s, PD, or in-service, was mandatory for all teachers, and it often took the form of a "convention" that occurred over multiple days, generally in the evening (Grosz, 2004). There was an increase in the number of students and the need for

additional teachers; however, many teachers were typically poorly educated and had gaps in the subject matter. During the 1900s through the 1940s, PD was a way to help remedy teachers' deficiencies and improve their academic competency (Grosz, 2004).

The first wave of school reform began in 1957 when the Russians launched the satellite Sputnik. The United States did not want to be academically behind the Russians. In reaction to Sputnik, policymakers focused on academic content, including math, science, and English (Grant et al., 2001). A majority of PD in the 1950s was in the form of workshops where teachers could collaborate with specialists, curriculum coordinators, or resource persons. The cornerstone of these workshops was to increase student achievement in the core academic subjects. In the 1960s, President Lyndon Johnson's War on Poverty legislation aimed to provide equal educational opportunities for low-income families (Grant et al., 2001). Consequently, the federal government has become increasingly more involved in building educational policies.

In 1965, the ESEA focused on improving equal access to education for schools with high percentages of students from low-income families. Title II allocates most funds to state and district-level activities to improve teacher and student learning, and Title II includes PD (Lin, 2013; McKnight, 2018). In 2001, NCLB reauthorized ESEA and introduced standardized testing to monitor student achievement in schools with low-income students (Okere, 2011). However, title II funds were still allocated for training, preparing, and recruiting high-quality teachers and principals. ESEA and NCLB caused educators to analyze the PD schools offer to meet the needs of educators and students (Grosz, 2004).

School reform continued with the publication of A Nation at Risk in 1983 (DoE, 2008). The United States educational system was not keeping the pace of progress as other nations. This report indicated the school system had lost sight of the purpose of education, become complacent, and needed to increase the expectations for educators. Twenty-five years later, in a review of a Nation at Risk, the report acknowledges that there is still much work to be done; however, we are a more informed nation (DoE, 2008). The report explained that we are at greater risk because of rising demands in the global economy, which requires students to be educated at a higher level to compete. Unfortunately, our education system is not keeping up with these growing demands (DoE, 2008). Haug and Mork (2021) explained that to prepare students for handling the complexity of global societies, policy documents, and educational reforms, they need to refine 21st century skills. Such skills and competencies include engaging in high-level reasoning, understanding content, and complex problem-solving skills.

California Teacher Standards

The CSTPs were developed in the 1990s and intended to provide new educators with a framework for formative assessment and best practices. The standards are organized into six interrelated domains of teaching practice. These standards include:

- Supporting and engaging learners.
- Effective learning environments.
- Subject matter knowledge.
- Designing learning experiences.
- Assessment.
- Developing as a professional educator.

These six areas allow educators to reflect on their practice and student learning, create professional goals to increase student learning, and assess their progress towards these goals (Commission on Teacher Credentialing, 2009).

Standard six addresses developing as a professional educator and identifies that teachers would reflect on their teaching practice to support student learning establish goals and engage in continuous professional growth, engage in collaboration, and learn how to employ families to help students. Standard six also addresses (a) enlisting local communities to help students; (b) managing professional responsibilities; and (c) demonstrating professional responsibility, integrity, and ethical conduct (Commission on Teacher Credentialing, 2009). The state of California recognizes the importance of ongoing PD for educators.

The CSTPs acknowledge that teachers are never finished with their professional education because they teach a diverse student population in a rapidly changing world (Commission on Teacher Credentialing, 2009). Therefore, high-quality PD is vital to increase educators' skills while helping meet the needs of 21st century learners. However, there are discrepancies between the teaching and learning educators provide their students and the teaching they experience with their professional learning (Hunzicker, 2011; Tate, 2012).

Theoretical Frameworks

The Russian psychologist, Lev Vygotsky, became noted for his research on learning and development. His work focused on the idea that most learning was due to social interactions, such as adult and child relationships (Vygotsky, 1978). Vygotsky's (1978) sociocultural approach to learning saw the process of learning as a whole

experience that is unique to each individual and situation (Chism, 2000). Effective PD must connect to the learner's perspective (Smith, 2016). As a learner, the socio-cultural context for the teacher is the school community, including other teachers, students, parents, and administrators. Learners work within a zone of proximal development (ZPD), learning through collaboration with more capable peers (Vygotsky, 1978). According to Roth and Lee (2007), ZPD can generally apply to learners. Many teachers participate in a community of practice, which serves as PD, where teachers learn from each other within a supportive framework (Francois van, 2017).

Adult Learning Theory

Knowles' theory of adult learning, known as andragogy, is based on several assumptions. Knowles et al. (2005) described these assumptions as adults needing to see a reason or purpose for the learning, be self-directed, and the learning needs to be task-centered. Educators are more likely to engage in professional learning when it has personal meaning and positively impacts their lives. Lutrick and Szabo (2012) found four themes regarding what instructional leaders viewed as effective PD traits, agreeing with Knowles' theory. These themes were that PD should be ongoing, collaborative, data-driven, and interest-driven in design. Educators will identify the value-added to their jobs when PD aligns with these themes, and they will often be more engaged in the PD. According to Disch (2020), many PD programs during the 1980s and 1990s failed to look at the process of adult learning. Effective PD shapes teachers' thinking, shaping their practice (Disch, 2020). Therefore, it is vital to incorporate adult learning theory into the design and delivery of PD programs so that educators can apply the learning to their practice.

Professional Development Models

Gissy (2010) explained PD could occur in various formats, including face-to-face meetings, informational sessions, weeklong academies, online meetings, and multiple publications. Additional PD formats included PD schools, coaching, and mentoring. A school site and district can determine PD. Some smaller school districts do not provide professional learning opportunities, and the teachers must pursue their professional learning outside of the school setting.

One PD model is professional development schools (PDS), which are specially structured schools where higher education institutions and public schools partner. PD improves schools and teachers and supports growth in researched-based areas of education (Gissy, 2010). According to the National Commission on Teaching & America's Future (1996) PDS improve teaching quality and student learning.

Workshops generally occur over a short period, are often offered by consultants outside of the school, and do not include follow-up or support. Guskey and Yoon (2009) described workshops as often criticized as ineffective and a waste of time and money. However, studies have indicated a positive relationship between PD and improvements in student learning after teachers attend workshops. Practical workshops included research-based instructional practices, involved active-learning experiences, and provided participants with opportunities to adapt the learning to their classrooms (Guskey & Yoon, 2009). The positive relationship between workshops and increased student achievement implies this PD format is a good choice for teachers, mainly because the workshops occur quickly.

Instructional coaches work from a school site or district level to provide instructional support for teachers. Coaching looks different based on the unique needs of each teacher. For example, coaches can offer in-class mentoring, teach model lessons, provide feedback, or share new ideas with teachers (Blackburn, 2020). There are also different methods of coaching. According to Blackburn (2020), embedded coaching occurs during instructional time and promotes collaboration. Okere (2011) advocates for peer coaching when teachers take charge of their learning and practice new classroom strategies with peers. Okere identified peer coaching as a method that results in school achievement.

The Changing Needs of the Teacher

Educators prepare students to contribute to the ever-changing and rapidly expanding global workforce. In 2006, the U.S. Department of Education (DoE) released a report entitled Answering the Challenge of a Changing World: Strengthening Education for the 21st Century. This report drew attention to the need for America to innovate and improve education while helping students develop the skills necessary to compete and succeed in higher education and the workforce. Employers are seeking employees who demonstrate 21st century skills, such as flexibility, creativity, critical thinking, and being knowledgeable in their area of expertise (DoE, 2006; World Economic Forum, 2020). To better prepare students for the world, teachers must continually pursue high quality, research-based PD (DoE, 2006).

The purpose of PD is to introduce teachers to strategies that help meet the changing needs of students (Guskey, 2009). PD programs designed well can inspire and motivate teachers to continue their lifelong learning (Potchka, 2009). Because today's

teachers must understand how to reach students from various backgrounds, learning opportunities must provide teachers with strategies for success (McCarthy & Riley, 2000; Potchka, 2009). PD is an essential part of an educator's personal growth (Bastin, 2003).

Statement of Research Problem

Educators are preparing students for college, career, and civic life, yet many lack the wide range of skills necessary to contribute to a rapidly changing workforce and society (Jimenez, 2020). A solid K-12 education is critical for the United States to thrive in a global economy (Klein & Rice, 2014). Professional learning is vital for increasing student achievement (Fullan, 2010). To better prepare students for the global workforce, PD must train teachers to integrate 21st century learning skills into daily educational routines to prepare graduates to be global contributors. District-level PD systems are responsible for preparing teachers, and a "one-size-fits-all" approach does not work.

Innovative PD systems can provide teachers with a mechanism to engage in personalized, collaborative, meaningful professional learning. Organizations employ PD administrators, staff development specialists, and instructional coaches to design and deliver ongoing professional learning opportunities. They are also responsible for providing access to high-quality PD for all teachers. According to Rodman (2018), effective PD plans should consider the learners' needs to be sustainable and ongoing. Contrary to a typical sit and get PD workshop model, PD should be instructional focused, job-embedded, and allow for teacher's voice and choice (Hunzicker, 2011; Rodman, 2018).

Although various research has been conducted on K-12 organizations and the benefits of PD, there is limited research on the systems approach of differentiated

professional learning opportunities in a school district. Models of PD can include collaboration with a coach or mentor, engaging in professional learning communities (PLC), or site-based learning, which is determined by the needs of individual teachers (Latz et al., 2009; Nishimura, 2014; Svendsen, 2020). However, studies about PD trends and models do not always include assessment measures that determine the effectiveness of professional learning and its impact on student achievement (Guskey & Yoon, 2009; McKnight, 2018). Additionally, the amount of research on the professional opinions of expert-status PD administrators on high-quality PD models for K-12 organizations is lacking.

Purpose Statement

The purpose of this futures Delphi study was to determine what experts in professional development predict the delivery and content of K-12 education will look like in 2026 and beyond, to determine how the experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted professional development in 2026 and beyond.

Research Questions

The following questions were investigated to address the purpose of the study:

- 1. What do experts in professional development predict the delivery and content of K-12 education professional development will look like in 2026 and beyond?
- 2. How do experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond?

- 3. How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for delivery and content of professional development?
- 4. How do experts in professional development respond to the predictions made from Research Question 3, and what final suggestions do they make for implementation in 2026 and beyond?

Significance of the Study

The DoE (2008) emphasized the importance of PD in educational reform throughout the years. Providing ongoing opportunities for differentiated PD for teachers will continue to positively impact student achievement (Guskey & Yoon, 2009; Yoon et al., 2007). This study is designed to supplement existing research on PD in K-12 organizations by identifying and outlining the most effective PD activities utilized in K-12 districts that have allowed PD administrators to reach expert status and subsequently impact student achievement. The outcomes of this study are significant to practitioners in the education field in various ways.

Studies of PD in K-12 organizations already point to the benefits of differentiated PD (John, 2014; Rodman, 2018). John (2014) determined that one-size-fits-all professional learning will not work for teachers. Instead, teachers need a variety of instructional strategies based on their grade level and content area to meet the diverse needs of their students. Nishimura (2014) expands on this notion, explaining that for meaningful change to occur, coaching, which is a model of PD, needs to be differentiated to be relevant to the needs and interests of the teacher.

This study will identify effective PD systems that improve teachers' skill sets as outlined in the Commission on Teacher Credentialing, (2009). As a result of this added research, district leadership would strategically create a more effective PD plan that systematically supports ongoing professional growth and allows teachers to progress toward reaching their personal learning goals.

Additionally, the outcomes of this study regarding effective PD systems will be of great significance to districts within the state of California that are underperforming, as indicated by the California Assessment of Student Performance and Progress (CAASPP). For example, Foster (2004) examined 30 California school districts that administered, scored, and analyzed common assessments in a way that guided PD and led to a change in teaching strategies. As a result, these districts saw growth in test scores on the statewide-standardized tests. The article *Shifting the Focus to Learning: California's Accountability Debates* (EdSource, 1998) explained that standards are rigorous, and teachers must acquire a deeper knowledge of subject matter and how to teach it. Subsequently, PD should be comprehensive to support the students' needs.

Aside from identifying best practices for sustainable and effective PD in a K-12 organization, this study will potentially inform PD models for district leaders' identification of research-based best practices and sustainable models of PD that meet the needs of all learners.

Definitions

Andragogy or adult learning theory. Framework for teaching adult learners. The core assumptions are that adult learners are self-directed, bring experiences to new

learning, demonstrate a readiness to learn, and learn with a purpose (Hunzicker, 2011; Knowles et al., 2005; Svendsen, 2020).

Differentiated professional development. Altered learning experiences through a change in the content, learning process, learning environment, and products created (John, 2014).

Instructional coaching. Instructional coaches partner with teachers to analyze current classroom practices and set goals. Then, the coach and teacher will identify instructional strategies that will help meet the goals, and the coach will provide support as needed (Latz et al., 2009).

Professional development (professional learning, staff development, teacher professional development, and teacher training). The process of educators learning to improve their practice. The new learning could occur in various formats, including faceto-face meetings, informational sessions, and multiple publications.

Professional learning community. A model of PD where educators work collaboratively to improve student learning (DuFour, 2004).

Delimitations

This study was delimited to expert K-12 PD administrators in California. One expert K-12 PD administrator was selected from 15 to make up the expert panel. Data was collected from this panel of experts during the timeframe of eight weeks.

Organization of the Study

This research study consists of five chapters, a reference list, and appendices.

Chapter II provides background and a review of the literature surrounding the history of K-12 education PD, educational policy, adult learning theory, PD models, and the

changing needs of the K-12 educational teacher. Chapter III outlines the research design and methodology, population, sample, instrumentation, and procedures used to collect and analyze data. Chapter IV presents the data collection results and analysis of the findings of this study. Chapter V contains the summary, findings, conclusions, and recommendations for further research.

CHAPTER II: REVIEW OF THE LITERATURE

Literature Review

The literature review explores the current research on PD in K-12 public schools and the different PD models available to educators. The chapter opens with an overview of the role of PD in K-12 education. Teacher PD is critical to increased student achievement (Guskey, 2002; Yoon et al., 2007; Zambak et al., 2017). Hattie (2012) argued that teachers are the most important factor influencing student achievement. According to Yoon et al. (2007), PD affects student achievement through enhanced teacher knowledge, skills, and improved teaching practices. This literature review also explores the history of K-12 PD and the evolution of teaching models.

Policymakers have acknowledged the importance of ongoing learning for educators, evident from the historical perspectives of K-12 PD. And so, this literature review will continue with a discussion of the impact of federal reforms on PD policies and the development of the California standards for the teaching profession. Educational reform has consistently acknowledged the need for ongoing professional learning, but the policies have never specifically identified how to provide effective PD. As such, each school district in California determines how to structure PD opportunities differently, spending the funds in various ways.

As the literature review continues, several different models for providing PD are analyzed. PD opportunities include conferences, workshops, coaching and mentoring, and PLCs. In addition, informal learning, including social media and blogs, has become a form of asynchronous PD. Consequently, the wide variety of PD models requires a deeper understanding of the effectiveness of the design and delivery of the chosen model.

Theories of adult learning enrich the design and delivery of PD, a critical component of the success of PD.

PD is continually evolving; however, little research has been done to identify strategies that bridge the gap from theory to practice in teacher PD. Determining the effective strategies when designing and delivering PD requires closely examining the connection between PD, high-quality teaching, and student achievement. The problem is that there are many challenges when determining the effectiveness of implementation strategies for PD because of the multiple variables contributing to PD. Few researchers are looking at assessing the effectiveness of PD in K-12 school districts and the impacts on increased student learning, even though PD is a critical component of student achievement. Thus, Chapter II will conclude with an overview of the gaps identified in the literature review, underlining the need for this research study. A synthesis matrix is included in Appendix A, which displays the alignment between the themes and literature.

The Role of Professional Development

Research uncovered a variety of definitions of teacher PD. Guskey (2002) defined PD as processes and activities designed to enhance educators' professional knowledge, skills, and attitudes. Hargreaves' (1995) definition of PD focused on the importance of purpose, passion, and desire in teaching. Guskey (1994) and Hargreaves (1995) concur that PD is an integral part of the teaching profession, and the journey should be meaningful and self-directed. Teachers refine their craft through ongoing, voluntary PD (Pokhrei & Behera, 2016). However, educators need access to structured opportunities for high-quality professional learning to remain current on advances in their field (Guskey, 1994; Svendsen, 2020).

High-quality PD leads to increased student achievement (Guskey, 2002; Yoon et al., 2007; Zambak et al., 2017). Characteristics of high-quality PD include coherence, active learning, sufficient duration, collective participation, and a focus on content knowledge (Darling-Hammond & McLaughlin, 1995; Yoon et al., 2007). In addition, teachers need to be actively involved in PD through collaboration, which includes observing one another, modeled instruction, and opportunities to discuss instructional problems and create solutions (Darling-Hammond & McLaughlin, 1995; Guskey, 2009; Svendsen, 2020). Although there is agreement that high-quality PD is necessary to increase student achievement, there is a shortage of such opportunities (Frerichs et al., 2018; Yoon et al., 2007). Guskey (2009) asserts that no improvement effort in the history of education has succeeded without the teachers' access to high-quality PD.

According to the National Commission on Teaching and America's Future (1996), report entitled *What Matters Most: Teaching for America's Future*, effective PD should not be a one-time experience; rather, it needs to provide ongoing, follow-up support. By interacting in PD continually, significant teacher development occurs (Svendsen, 2020). Therefore, regardless of how K-12 educational organizations are structured, PD is fundamental to increasing teacher effectiveness and student achievement.

The Commission on Teacher Credentialing (2009) explained that a high-quality educator matters the most for school student development and learning. Effective teaching requires extensive content knowledge, passion, and commitment (Commission on Teacher Credentialing, 2009; Hargreaves, 1995). Effective teachers also strive to know and understand their students, families, communities, and the individual strengths

and needs of the students. In addition, effective teaching requires thoughtful planning focused on standards-based defined learning objectives (Commission on Teacher Credentialing, 2009). Finally, effective teachers actively engage within a professional learning community (Commission on Teacher Credentialing, 2009; National Commission on Teaching and America's Future [NCTAF], 2006). PD opportunities contribute to high-quality teaching and learning experiences (Commission on Teacher Credentialing, 2009; DoE, 2008; Yoon et al., 2007).

Historical Perspective of K-12 Professional Development History of Professional Development: 1800s to 1960s

The methods for planning and providing PD have dramatically changed. In the 1800s, PD, or in-service, was mandatory for all teachers. It often took the form of a "convention" that occurred over multiple days, generally in the evening (Neil, 1986). There was an increase in the number of students and the need for additional teachers; however, many teachers were typically poorly educated and had gaps in the subject matter they were teaching. During the 1900s to 1940s, PD was a way to help remedy teachers' deficiencies and improve their academic competency (Neil, 1986).

The first wave of school reform began in 1957 when the Russians launched the satellite, Sputnik. The United States did not want to be academically behind the Russians. In reaction to Sputnik, policymakers insisted on a focus on academic content, including math, science, and English (Grant et al., 2001). Most PD in the 1950s was in the form of workshops where teachers could collaborate with specialists, curriculum coordinators, or resource persons. The cornerstone of these workshops was to increase student achievement in the core academic subjects. In the 1960s, President Lyndon Johnson's

War on Poverty legislation aimed to provide equal educational opportunities for low-income families (Grant et al., 2001). In addition, the federal government had become more involved in building educational policies.

In 1965, the ESEA focused on improving equal access to education for schools with high percentages of students from low-income families. Teacher PD was a targeted area of the act. In 2001, NCLB reauthorized ESEA and introduced standardized testing to monitor student achievement in schools with low-income students. In addition, Title II funds were allocated to train, prepare, and recruit high-quality teachers and principals. ESEA and NCLB raised the standards and accountability, which caused administrators to analyze the PD offered to meet better the needs of educators and students (DoE, 2006).

History of Professional Development: 1970s to Present

In the 1970s, B. F. Skinner's learning theory, behaviorism, began a shift in the classroom that required additional professional learning opportunities for teachers. Behaviorism follows key principles, including how behavior is learned, the importance of immediate feedback, and effective teaching, including positive reinforcements (Al-Shammari et al., 2019; Grant et al., 2000; Skinner, 1968). School reform continued with the publication of A Nation at Risk in 1983 (DoE, 2008). This report indicated the school system had lost sight of the purpose of education, become complacent, and needed to increase the expectations for educators. Reform initiatives have challenged educators with new skills and responsibilities, which has required a change in practice, which occurs through ongoing PD (Corcoran, 1995).

In the 1980s, Madeline Hunter's curriculum planning model fueled the continued educational reform. Teachers attended PD through workshops on effective teaching

(Grant et al., 2001). According to Grant et al. (2001), the Hunter model trained teachers in lesson design and delivery components. Many teacher evaluations were based on the successful implementation of the model (Grant et al., 2001). Ongoing PD is a critical component of implementing and sustaining educational reform because continual PD allows teachers to deepen their content knowledge and learn new methods of teaching (Corcoran, 1995).

Because PD is recognized as necessary, several organizations have developed guidelines and standards to direct ongoing teacher education. The standards of the DoE and the National Staff Development Council (NSDC) apply to all general PD (Grant et al., 2001). NSDC standards address three domains: (a) context, (b) process, and (c) content. The NSDC ensures student success through staff development and school improvement. Hirsh (2007) stated more than 40 states had adopted PD standards, and more than 25 are using NSDC's standards. However, PD assessment and its implementation and impact on student achievement is still inconsistent across the nation. Significant resources have been spent on funding, planning, and implementing PD. Unfortunately, the NSDC does not have an evaluation tool to assess the effectiveness of the standards, which allows for inconsistency in measuring effectiveness. PD should enrich teachers' knowledge of content and pedagogy and increase student achievement (Guskey, 2002; Pharis et al., 2019; Yoon et al., 2007). Creating meaningful and effective PD for teachers is an ongoing challenge.

The Impact of Federal Reforms on Professional Development Policies

Teacher PD is an essential aspect of growth as an effective educator (Guskey, 1994). As a result, most educational reform initiatives acknowledged the importance of

PD and included financial allocations that contribute to PD implementation (Corcoran, 1995; Guskey, 1994). The work complements the vital work at the state and federal levels (DoE, 2008). However, education reforms do not dictate how PD must occur or what topics will be taught. According to Guskey (1994), because of the variability, it is difficult to assess the effectiveness of PD as a result of educational reform.

In 1965, the ESEA targeted teacher PD, and in 2001, NCLB reauthorized ESEA. However, PD was still an area of focus for educational reform because of the increased expectations for students and teachers (Corcoran, 1995). The DoE (2008) published *A Nation Accountable, Twenty-Five Years After a Nation at Risk*. The report stated in 1989, President G. W. Bush convened with the nation's governors and agreed to adopt national K-12 performance goals for the year 2000. The DoE (2008) explained that since the implementation of the national performance goals, some states have tried to align their teacher training with the goals, standards, and assessment.

The educational reforms require high-quality PD, and research recognizes the short supply (Corcoran, 1995; Yoon et al., 2007). The NCLB mandated teachers receive high-quality learning opportunities and set five criteria for PD to be considered high quality:

- PD must first be sustained, intensive, and content-focused.
- Second, PD must be aligned and directly related to state academic content standards, student achievement, and assessment.
- Third, PD improves and increases teachers' knowledge of the subjects they teach.
- Next, it is grounded in research-based practices.

• Finally, PD is regularly evaluated for the effects on teacher effectiveness and student achievement (Yoon et al., 2007).

The current standards and accountability movements in educational reform have forced K-12 educational organizations to analyze the PD their teachers have access to. The educational reform has also provided reliable data to evaluate student performance and determine areas teachers need additional PD to target areas where increased student achievement must occur (DoE, 2008). As a result, the educational system has access to valuable data, coupled with high-quality PD, and will continue to increase student achievement.

California Standards for the Teaching Profession

The CSTP provide a common language and definitions of teaching for educators as they develop their practice (Commission on Teacher Credentialing, 2009). The CSTPs include six standards encompassing:

- Engaging and supporting learners.
- Learning environments.
- Organizing subject matter.
- Lesson design.
- Assessment.
- Developing as a professional educator (Commission on Teacher Credentialing, 2009).

The CSTPs is the cornerstone of teaching policy in California, and many districts across the state use these standards for teacher evaluation (Whittaker et al., 2001).

Standard six addresses developing as a professional educator and states that teachers will reflect on their teaching practice to support student learning, establish goals and engage in continuous professional growth, collaborate, and learn how to employ families to support students. Standard six also addresses enlisting local communities to help students, managing professional responsibilities, and developing in the area of demonstrating professional commitment, integrity, and ethical conduct (Commission on Teacher Credentialing, 2009). The state of California recognizes the importance of ongoing PD for educators.

From 1988 to 1992, according to Whittaker et al. (2001), the California New Teacher Project work revealed the need for a commonly understood set of expectations by beginning teachers. The framework was developed and outlined these expectations, including the knowledge, skills, and abilities required by beginning teachers. The framework also acknowledged that teaching is complex and requires time to develop teaching expertise (Whittaker et al., 2001). The framework's goal was to create the most significant teaching elements rather than a checklist of behaviors (Whittaker et al., 2001). In 1997, the California Standards for the Teaching Profession were adopted, a milestone in the long-term effort to foster professionalism in California teaching (Whittaker et al., 2001).

The California Department of Education (CDE) (n.d.) includes professional learning in its framework for high-quality learning. The CDE (n.d.) provides opportunities for educators to engage in professional growth and help students learn. The CDE defines professional learning opportunities like workshops and other more traditional types of PD and elaborates with the following:

But it also goes further, engaging educators in self-reflection, peer support, experimentation, and modification of instruction and management practices based on student performance data, student work, and both learning and social behaviors. Through an intensive process of collaborative and job-embedded learning, educators can gain more than content knowledge or technical strategies-they can gain an improved understanding of their own teaching and learning and of the various ways by which students learn. Through this effort, educators also come together as a community of self-developing practitioners. (para. 1)

California recognizes the importance of professional growth and educators' impact on student achievement, yet there is no standardized assessment for the quality of professional learning. High-quality PD is essential to increase the skills educators possess, which implies assessing the quality of PD and its impact on student achievement.

Professional Development Effective Practices

There is growing evidence about what defines effective PD practices. Miles and Guiney (2000) explained the importance of shared goals within districts and schools when designing teacher PD opportunities. PD is more effective when the effort is district-wide, versus varying by school and teacher and aligned to common district goals (Corcoran, 1995; Darling-Hammond & McLaughlin, 1995; Miles & Guiney, 2000). Successful PD must be high-quality and relevant to teachers' needs (Corcoran, 1995; Elmore, 2004). Continued follow-up support is important in effective PD (Darling-Hammond & McLaughlin, 1995; Guskey, 1994). When school districts align the PD

offered to district goals and teachers' needs, student learning will increase (Corcoran, 1995; Miles & Guiney, 2000; Yoon et al., 2007; Zambak et al., 2017).

Professional learning includes various experiences that deepen teachers' content knowledge and add to the methods of teaching used in their classroom. These experiences can consist of collaboration with colleagues and opportunities for reflection (Corcoran, 1995; Frerichs et al., 2018; Guskey, 2009). One of the most effective PD practices is participation in a PLC (DuFour, 2004; Miles & Guiney, 2000; Nishimura, 2014; Showers & Joyce, 1987). According to Corcoran (1995) and Guskey (2009), good PD should address curriculum content and design and instructional and assessment strategies that allow students to engage in higher-order thinking. For example, the constructivist teaching model allows teachers the time to explore, question, and debate new learning and ideas (Corcoran, 1995). Engaging in this process helps teachers grow as professionals, apply their learning to instructional routines, and master the new content through collaboration and practice.

Guskey (2009) determined that effective PD may not be found on a list of "best practices." Still, instead, core elements of effective PD can be adapted to meet the needs of a particular district or school. Darling-Hammond and McLaughlin (1995) stated that district policies directly affect teacher PD opportunities. Studies have found that when districts and schools value lifelong learning and create a culture of collaboration, the benefits can include improved classroom instruction and enhanced student learning (Archambault et al., 2010; John, 2014). Additionally, strategies included among the consensus in the literature regarding effective teaching and learning elements focus on student and teacher learning, content and pedagogy, opportunities for collaboration and

reflection, and sustained support (Corcoran, 1995; Darling-Hammond & McLaughlin, 1995; Haug & Mork, 2021).

Professional Development Delivery Models

PD can be delivered in various models, which the school district or outside vendors can provide. Also, PD content can be determined at district, school, or teacher levels. Some smaller districts do not offer professional learning opportunities, and the teachers must pursue their professional learning outside the school setting; however, the most effective PD aligns with the district's goals (Miles & Guiney, 2000). PD opportunities include conferences, workshops, coaching and mentoring, and PLCs. In addition, informal learning, such as from Twitter and blogs, has become a form of asynchronous PD.

Workshops and Conferences

Workshops and conferences generally occur over a short period and are often offered by consultants during the school day or on the weekend. Guskey and Yoon (2009) described workshops as often criticized as ineffective and a waste of time and money. Experts from outside the school typically address trending topics and issues, leaving the attendees with some practical tips, but they seldom offer follow-up support (Corcoran, 1995). Teachers may leave a workshop with handouts and ideas; however, they do not gain a deep understanding of the new learning in one short PD workshop (Corcoran, 1995). On-going PD is needed to better support teachers in implementing new learning and best practices (Corcoran, 1995; Pokhrei & Behera, 2016).

Conversely, some studies have indicated a positive relationship between PD and improvements in student learning after teachers attend workshops. Effective workshops

included research-based instructional practices, involved active-learning experiences, and provided participants with opportunities to adapt the teaching to their classrooms (Guskey & Yoon, 2009). The positive relationship between workshops and increased student achievement implies this PD format could be a good choice for teachers, mainly because the workshops occur over a short time.

District Provided Learning Opportunities

An effective method of providing PD is building capacity from a district level by making principals be the lead learner and providing teacher learning opportunities aligned with district goals (DuFour, 2004; Luke & McArdle, 2009; Sharratt & Fullan, 2009). Anderson (2006) also reviewed research on district effectiveness and identified district-wide, job-embedded PD as a critical component to success. Sharratt and Fullan (2009) elaborated that administrators have a shared responsibility and must engage district PD experts to facilitate active professional learning opportunities based on teacher or student needs. According to Darling-Hammond and McLaughlin (1995), district leadership must provide schools with the necessary resources to support PD and increase student learning.

School leaders are expected to play a big role in improving teaching and learning. Tong and Razniak (2017) explained that administrators must understand their staffs' needs and strengths. Administrators can further support teachers by providing opportunities for collaboration and reflection (Tong & Razniak, 2017). Building a lifelong learning and risk-taking culture are key components administrators can offer their teachers (Corcoran, 1995; Darling-Hammond & McLaughlin, 1995; Tong & Razniak, 2017). When teachers feel safe, they will be more willing to apply the new

learning in the classroom; therefore collaborative leadership is a key to providing opportunities for PD at the district and school levels.

In 1998, San Diego City Schools engaged in a three-year educational reform process, which focused on increasing student achievement by supporting teaching and learning in the classroom. Hightower (2002) stated that San Diego City Schools recognized that school districts can influence teacher learning by engaging the principals as instructional leaders. They restructured their district office and created a team of seven principals who became district-wide "instructional leaders" and created seven working groups of 25 principles each, called "learning communities." These teams engaged in long-term, professional learning networks for teachers and principals, collaborated, and provided opportunities for reflection and refinement of practice (Hightower, 2002). Continuous PD was a cornerstone in San Diego City School's reform agenda. The district also created a network of highly qualified teachers who served as "peer coaches" or "staff developers." Each school site in the district had one of these coaches, and they would focus on coaching teachers in research-based strategies for learning within the school context. Historically, district PD models can include PLCs, instructional coaching, and content-specific workshops (DuFour, 2004), which were incorporated into San Diego City School's reform agenda. System changes occur over time, and San Diego City School' is an example of a large school district that successfully implemented change focused on increasing student achievement by supporting teaching and learning in the classroom (Hightower, 2002).

Coaching Model

PD is designed to improve schools and teachers, thereby increasing student achievement by learning through researched-based areas of education (Guskey, 2002; Yoon et al., 2007; Zambak et al., 2017). Novota (2003) indicated that when PD has been built into the daily teaching job, it has changed teacher practice and increased student learning. Teachers can improve their practice through observation, reflection, and coaching (Corcoran, 1995; Darling-Hammond & McLaughlin, 1995; Farr & Saltmarsh, 2018; Les, 2013). Instructional coaches work from a school or district level to provide instructional support for teachers. Frerichs et al. (2018) explained that coaching sessions are based on the unique needs of each teacher, which allows the coaching to be relevant and personalized. Coaches can provide in-class mentoring, teach model lessons, provide feedback, or share new ideas with teachers (Blackburn, 2020). Coaching is to remain supportive, and the coach does not function as an evaluator (Frerichs et al., 2018; Knight; 2019; Les, 2013).

There are different coaching methods, such as using the impact cycle, developed by Jim Knight. Coaching is an intentional process with three stages: identify, learn, and improve (Knight, 2019). Knight (2019) explained that during the identify stage the teacher and coach partner to determine the current reality of the classroom, a goal, and a strategy that can be implemented while working towards achieving the goal. Then, during the learn stage, the coach will model instructional strategies while also encouraging the teacher to adjust the strategy based on the needs of their students. Finally, during the improve stage the teachers implement the strategies and make adaptations until the goal is met. Farr and Saltmarsh (2018) described the impact cycle as a partnership approach to

instructional coaching, including a collaborative conversation after each instructional cycle. Knight asserts that the instructional coaching partnership improves teaching and positively impacts student learning.

Another type of coaching is peer coaching when a teacher-leader helps a peer improve their instruction by engaging students in 21st century learning activities (Les, 2013). Coaches help teachers take charge of their learning while practicing new classroom strategies. A peer coach might provide just-in-time training or resources, coplanning learning activities, modeling or team teaching to demonstrate effective teaching and reflection afterward, and observing teachers and reflecting on what was observed (Darling-Hammond & McLaughlin, 1995; Les, 2013). Les (2013) discovered that the process of observation and reflection is the most effective form of formative assessment for teachers, and it is a key to lifelong learning. Peer coaching is a research-based method of PD that results in helping teachers improve student learning (Darling-Hammond & McLaughlin, 1995; Les, 2013).

Blackburn (2020) states that embedded coaching occurs during instructional time and promotes collaboration. Instructional coaching provides differentiated PD for teachers because the coaching styles and models may vary depending on the teacher's needs (Les, 2013; Weidenfeld & Bashevis, 2013). However, a disadvantage of coaching is the required investment of time and energy on the part of the school to launch and maintain effective coaching programs. Also, some coaching models need the coach to be an expert in the content knowledge, and it can be challenging to find experts willing to leave the classroom and coach others (Blackburn, 2020; Les, 2013). Knight (2019) described additional skills needed for effective coaching, including (a) discipline, (b)

organization, (c) professionalism, (d) flexibility, and (e) good listening skills.

Nevertheless, instructional coaching is an effective form of job-embedded professional learning that requires collaboration and increases student learning, even though there can be implementation challenges.

Professional Learning Community

A PLC is a school, grade-level team, high school department, or an entire district that moves away from an industrial model of education and enables a new model, which allows the team to function as a learning organization to improve student achievement (Beach, 2012; DuFour & Eaker, 2009). In a PLC, one must focus on the learning rather than the teaching, engage in collaboration, and be accountable for the results (DuFour & Eaker, 2009). DuFour (2004) further explained that a PLC model requires schools to have a set of characteristics, such as (a) a shared mission, vision, and values; (b) collective inquiry; (c) collaborative teams; (d) an orientation toward action and a willingness to experiment; (e) commitment to continuous improvement; and (f) a focus on results. A vital component of a successful PLC is developing a culture of collaboration.

A culture of collaboration can be created when a PLC recognizes that they must work together to achieve their common goals (DuFour & Eaker, 2009). Corcoran (1995) added that teacher networks positively affect teacher motivation, subject matter knowledge, risk-taking, and overall commitment to improvement. An effective PLC builds structures to promote a collaborative culture because the teachers recognize that they must work together to achieve learning for all students (DuFour, 2004). Many PLCs face the challenge of shifting teachers' mindsets to one of collaboration versus working in isolation (DuFour, 2004; John, 2014). In addition, John (2014) contends that a necessary

prerequisite for breaking down teacher isolation is to embed time during the workday for collaboration, lesson planning, and assessing student work. Teachers must meet regularly to build trust, engage in productive collaboration, and focus on the results (DuFour, 2004; John, 2014; Tong & Razniak, 2017).

According to DuFour (2004), PLCs evaluate their effectiveness based on student achievement results. PLCs participate in the ongoing evaluation process, which identifies the current level of student achievement, goal setting, collaboration to achieve the goal, and then engaging in dialogue to analyze student data and determine the evidence of progress (DuFour, 2004). Teachers build social capital while interacting and engaging in conversation to improve teaching practices, therefore, student achievement (Tong & Razniak, 2017). For example, when teacher teams develop common formative assessments and participate in the evaluation process, they can determine the effectiveness of the teaching strategies implemented and replicate or make corrections as needed (DuFour, 2004). PLCs focus on continual improvement and results, which requires educators to embrace the process and foster a culture of collaboration.

Professional Learning Networks

Traditional PD models typically include courses offered by the school, district, university, or for-profit vendors. Most traditional PD is "one-size-fits-all" and "sit-andget," which does not meet the needs of the diverse educators attending the learning opportunities (Archambault et al., 2010; Dede, 2006; Tate, 2012). Therefore, many 21st century educators are forming PLNs, which is defined as a system of interpersonal connections and relationships that support learning (Beach, 2012; Krutka et al., 2017; Trust, 2012). PLNs often occur through social media and include informal learning

opportunities, which allows the learning to be self-directed, voluntary, connected, and collaborative (Archambault et al., 2010; Krutka et al., 2017). Additional benefits include immediate access to information and learning opportunities, diverse audiences, multimodal learning and communication, and immediate feedback (Archambault et al., 2010). PLNs are alternatives to traditional PD, allowing educators to take ownership of their learning by engaging in collaborative and asynchronous PD.

Common asynchronous online PLNs include blogs, Twitter, Wikis, podcasts, and online videos (Archambault et al., 2010; Beach, 2012). These platforms allow educators to acquire and share ideas and resources to enrich the learning experience. Reflection is essential when participating in a PLN because it enables learners to think critically about their instruction, beliefs, and student outcomes (Beach, 2012; Darling-Hammond & McLaughlin, 1995). Beach (2012) elaborates that for teachers to grow as professionals, they have to engage in reflection as they try out new ideas and practices. Unfortunately, a disadvantage of PLNs is the difficulty of tracking professional learning or quantifying the hours of professional education (Darling-Hammond & McLaughlin, 1995). Krutka et al. (2017) explained that learning through technology is still relatively new and evolving. Therefore, educators must be aware, reflective, and intentional about using technology for PD.

Barriers to Effective Professional Development

Educational reform initiatives, including the Race to the Top grant application (DoE, 2017) and the School Improvement Fund regulations (DoE, 2018), acknowledge the importance of PD and, as a result, include funding to support ongoing learning opportunities. Guskey (2009) explained that high-quality PD must be well organized,

carefully structured, and purposefully directed. Haug and Mork (2021) agreed, adding that high-quality PD allows teachers to engage actively, collaborate, and reflect on their learning. Barriers to effective PD include finding the time to provide professional learning, applying the new learning and reflecting on the practice (Guskey, 1994, 2009).

Professional learning should not occur in one-time workshops; instead, the learning should be extended over time and linked to classroom teaching (Guskey, 1994; John, 2014). In addition, John (2014) acknowledged the importance of time for individual and collaborative reflection when considering instructional changes resulting in new learning from PD. New strategies acquired during PD also take time to implement in the classroom and gather data (John, 2014). Guskey (1994) and John (2014) assert that teachers are more likely to try out the new practices, receive feedback, and reflect when learning activities are extended over a long time. One strategy to support the time barrier is providing job-embedded PD opportunities.

PD greatly influences job-embedded and built into the regular workday (Darling-Hammond & McLaughlin, 1995; John, 2014; Tate, 2012). Job-embedded PD refers to teacher learning with a foundation in daily teaching practices and is designed to enhance teachers' content-specific instructional practices (Darling-Hammond & McLaughlin, 1995). Job-embedded PD includes instructional coaching, mentoring, model lessons, and PLCs. One barrier to implementing job-embedded professional learning opportunities is the financial impact and access to human capital at a district and school level (Darling-Hammond & McLaughlin, 1995; Graham et al., 2020; Miles & Guiney, 2000). According to Miles and Guiney (2000), few districts are prepared to support teachers with these opportunities despite the substantial need for job-embedded professional learning. Most

PD offered does not offer ongoing support, is fragmented, and is one-size-fits-all (Corcoran, 1994; Hunziker, 2001; Miles & Guiney, 2000). District leadership is a critical component of implementing effective PD.

Darling-Hammond and McLaughlin (1995) explained the importance of district and site leadership when implementing high-quality PD that increases student learning. Fullan (2016) added, acknowledging the difficulty of improving whole systems, stating that districts often identify the wrong policy drivers, such as testing and evaluation. Fullan asserts success in improving systems lies in changing the culture and relationship toward the issues. Additional research agrees, explaining administrative leadership must create and sustain an environment where teachers feel safe to engage in new learning and risk-taking, and unfortunately, this is often ignored (Corcoran, 1995; Darling-Hammond & McLaughlin, 1995; Tong & Razniak, 2017). Organizational leadership plays a vital role in system improvement.

Challenges for Determining the Effectiveness of Professional Development

Blank et al. (2009) reported that current educational policies prioritize improving teacher quality and effectiveness, and PD can enhance the quality of teachers. However, according to Guskey (2009), determining the effectiveness of PD is complicated because studies can consume considerable time and resources, there are many variables, and it is often difficult to attract participants to the study. Another challenge for determining the effectiveness of PD is isolating the effects of the PD because schools often participate in multiple new initiatives (Guskey, 2009). Despite the challenges, gathering data and assessing PDs effectiveness is still critical.

Guskey and Yoon (2009) discussed the importance of gathering data on the effectiveness of PD in their study of *What Works in Professional Development*. They stressed the importance of planning and collecting data after PD. PD will not improve student achievement immediately and it takes time to see the impacts (Guskey, 2009; Yoon et al., 2007). Unfortunately, some PD leaders reluctantly collect data and assess their effectiveness (Guskey, 2009).

Guskey and Yoon (2009) explained that effective PD requirements include carefully structured, purposefully directed, focused on content and pedagogy, and organized time. Additionally, assessing effective PD is complex because it is unclear who needs to be assessed for effectiveness. Since a variety of people contribute to PD, including the designer, deliverer, and participant who applies the new learning in the classroom, there are many challenges in determining the effectiveness of PD (Guskey & Yoon, 2009).

Adult Learning Theory

Knowles' theory of adult learning, known as andragogy, is based on several assumptions, which vary from the assumptions of pedagogy, which is teaching children. Knowles et al. (2005) described these assumptions as adults needing to see a reason or purpose for the learning, be self-directed, and the learning needs to be task-centered. In addition, educators are more likely to engage in professional learning when the learning is differentiated, has personal meaning, and will positively impact their life (Knowles, 1984; Tong & Razniak, 2017).

Lutrick and Szabo (2012) found four themes regarding what instructional leaders viewed as traits of effective PD, which agrees with Knowles' theory. These themes were

that (a) PD should be ongoing, (b) collaborative, (c) data-driven, and (d) interest-driven in design. Educators can identify the value added to their jobs when PD is aligned with these themes. Unfortunately, not all PD applies the adult learning theory assumptions to the design or delivery of PD.

According to Southerland et al. (2016), many PD programs during the 1980s and 1990s failed to even look at the process of adult learning. Knowles (1973) referred to the traditional educational system as "progressively regressive" because the methods and approaches to adult learning had not progressed in innovations and education. Progress has been made in applying adult learning theory, and effective PD now shapes teachers' thinking, shaping their practice (Frerichs et al., 2018; Southerland et al., 2016). Frerichs et al. (2018) indicated adult learners require ongoing PD experiences to improve their skills and abilities. When designing adult learning experiences, build opportunities to collaborate and reflect, immediate options to practice, and develop a learning community (Frerichs et al., 2018; Tong & Razniak, 2017). Adult learning must shift from the teachers being passive participants to becoming active learners (Svendsen, 2020). Tong & Razniak (2017) assert the importance of incorporating adult learning theory into the design and delivery of PD programs so that educators are engaged and able to apply the learning to their practice.

21st Century Learning

Education aims to increase student learning and prepare students to contribute to an ever-changing society (Carbaugh et al., 2015; Vadeboncoeur, 2005). The Commission on Teacher Credentialing (2009) purports that today's diverse student population is rapidly evolving, and teachers need ongoing PD opportunities to foster excellence in

teaching and learning. PD is critical for increased student achievement (Yoon et al., 2007). To better prepare students to be contributors to society, educators must provide ample opportunities to apply 21st century learning skills such as communication, collaboration, critical thinking, and creativity in their daily instructional experiences. PD should prepare teachers with the strategies to engage students in these learning experiences where they apply 21st century skills; however, most PD falls short of this goal.

According to Darling-Hammond McLaughlin (1995), all systems of PD should be flexible and able to respond to the changing needs of educators. Darling-Hammond and McLaughlin argued policymakers need to be mindful of the demands of society, acknowledging when systems may need revising because of the changes in society. PD systems should be adaptable because structures in one school may not work in another. In addition, PD systems must provide multiple opportunities for collaboration and critical reflection for teachers to learn (Darling-Hammond & McLaughlin, 1995). Teaching is not a static occupation, and Carbaugh et al. (2015) further explained that rapid technological advances put additional pressure on educators to improve continuously.

The demands of the 21st century require continuous improvement to increase student learning. The Commission on Teacher Credentialing (2009) expanded on the notion that teaching is more than methodology. It is understanding student development, families, and communities, subject matter, instructional methods, and assessment of student learning, which is evolving. Administrators at the district and school levels are critical to supporting the improvement of teaching and learning (Carbaugh et al., 2015). Tong and Razniak (2017) indicated that administrators must develop a culture of trust

and risk-taking to foster 21st century learning. Teachers will be more willing to adjust their teaching practices if administrators create a culture of trust, promote collaboration, and model risk-taking (Corcoran, 1995; Novota, 2003; Tong & Razniak, 2017).

Leadership is vital in creating a safe learning environment for teachers to refine, develop, and share new learning strategies and better prepare students for the 21st century (Tong & Razniak, 2017).

Research Gap

While there is research that identifies the need for ongoing, job-embedded, PD that also includes opportunities for collaboration and reflection to prepare students with the skills needed for the 21st century and successfully implement the practices in the classroom, there is a need for more (Holme, 2019; Yoon et al. 2007). A historical review and synthesis of the existing literature surrounding the topics above lead to a gap in the research surrounding the impact of effective PD systems in K-12 districts and bridge the gap between theory and practice in organizations.

This gap points to the need to conduct a study using the Delphi methodology to identify what specific PD activities experienced by expert PD administrators in California helped them reach expert status and impact the desired teacher learning outcomes defined earlier. Secondly, the gap infers a need to study which activities are the most effective toward the same result. Lastly, the gap implies the need to allow the expert PD administrators to best implement these practical activities in future PD initiatives, including the design and delivery. A study created to answer these questions will help provide answers to school districts seeking strategies to help their district and school

reach expert status and subsequently positively increase student achievement through improved teaching and learning.

Summary

Teacher PD is an integral part of the teaching profession and is defined as processes and purposeful activities designed to enhance educators' professional knowledge, skills, and attitudes (Guskey, 2002; Hargreaves, 1995). High-quality PD leads to increased student achievement (Guskey, 2002; Yoon et al., 2007; Zambak et al., 2017). Characteristics of high-quality PD include coherence, active learning, sufficient duration, collective participation, and a focus on content knowledge (Darling-Hammond & McLaughlin, 1995; Yoon et al., 2007). Although there is agreement that high-quality PD is necessary to increase student achievement, there is a shortage of such opportunities, even though PD has evolved since the 1800s (Frerichs et al., 2018; Yoon et al., 2007).

Education reform initiatives have challenged educators with new skills and responsibilities, which has required a change in practice, which occurs through ongoing PD (Corcoran, 1995). In 2001, NCLB reauthorized the ESEA and introduced standardized testing to monitor student achievement in schools with low-income students. In addition, Title II funds were allocated to train, prepare, and recruit high-quality teachers and principals. ESEA and NCLB raised the standards and accountability, which caused administrators to analyze the PD offered to better meet the needs of educators and students (DoE, 2006). However, education reform allocates funds to provide ongoing learning for educators but does not dictate how PD must occur, which causes variability.

As a result, it is difficult to assess the effectiveness of PD and its impact on student achievement.

The research identified that PD is more effective when the effort is district-wide, versus varying by school and teacher and aligned to common district goals (Corcoran, 1995; Darling-Hammond & McLaughlin, 1995; Miles & Guiney, 2000). Also, successful PD must be high quality and relevant to teachers' needs (Corcoran, 1995; Elmore, 2004). PD workshops and conferences positively impact teachers; however, job-embedded learning opportunities that offer ongoing support are the most effective (Corcoran, 1994; Hunziker, 2001; Miles & Guiney, 2000). Unfortunately, some PD lacks active participation, collaboration, and reflection opportunities. Therefore, determining the effectiveness of PD is crucial when district leaders are making decisions on which models to design, deliver and provide teachers in the district.

Research studies on the effectiveness of PD can consume considerable time and resources, there are many variables, and it is often difficult to attract participants to the study. Guskey and Yoon (2009) emphasized the importance of collecting PD data so that data-driven decision-making can occur within school districts. Another factor affecting teacher PD is the integration of adult learning theory based on several assumptions.

Knowles et al. (2005) described these assumptions as adults needing to see a reason or purpose for the learning, be self-directed, and the learning needs to be task-centered.

When designing and delivering PD, facilitators should apply adult learning theory to prepare teachers with the strategies required to engage students in learning experiences where they use 21st century skills.

Research acknowledges administrators at the district and school levels are critical to supporting the improvement of teaching and learning by developing a culture of trust and risk-taking to foster 21st century learning (Carbaugh et al., 2015; Tong & Razniak, 2017). In addition, teachers will be more willing to adjust their teaching practices if leadership engages in behaviors that support and promote collaboration and model risk-taking (Corcoran, 1995; Novota, 2003; Tong & Razniak, 2017). And so, the purpose of this Delphi study is to determine what experts in PD predict the delivery and content of K-12 education will look like and how districts need to structure themselves to deliver the predicted content. This research may advance the theory of K-12 organizational leadership and contribute to PD literature and best practices.

CHAPTER III: METHODOLOGY

Overview

This study adds to the body of knowledge regarding K-12 educational PD, specifically determining what experts predict the delivery and content of K-12 education PD and how K-12 educational organizations will need to structure themselves to deliver predicted PD. This chapter describes the study's framework, beginning with the purpose statement followed by the research questions and research design. The population, target population, and sample utilized for this study are also addressed. Next, the study's instrumentation, including the process and procedures used to collect and analyze data, are examined. Finally, Chapter III concludes with the study's limitations and a summary.

Purpose Statement

The purpose of this futures Delphi study was to determine what experts in professional development predict the delivery and content of K-12 education will look like in 2026 and beyond, to determine how the experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted professional development in 2026 and beyond.

Research Questions

The following questions were investigated to address the purpose of the study:

1. What do experts in professional development predict the delivery and content of K-12 education professional development will look like in 2026 and beyond?

- 2. How do experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond?
- 3. How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for delivery and content of professional development?
- 4. How do experts in professional development respond to the predictions made from Research Question 3 and what final suggestions do they make for implementation in 2026 and beyond?

Research Design

This study used the classical Delphi method to collect data from K-12 administrators regarding the delivery and content of PD in K-12 educational organizations. Furthermore, the Delphi method allowed the researcher to determine how K-12 educational organizations in California will need to structure themselves to deliver predicted PD in 2026 and beyond. The RAND Corporation developed the Delphi method in 1950 (Dalkey & Helmer, 1963). The classical Delphi method is a forecasting process framework that uses various rounds of questionnaires sent to a panel of experts (Dalkey & Helmer, 1963; Sitlington & Coetzer, 2015). Using the classical Delphi method as a qualitative research design, the researcher obtained descriptive data through the questionnaires.

This study used the classical Delphi method to gather perceptual data from an expert panel of K-12 administrators who decide on PD structures in their districts. This methodology systematically collects information from a group of experts and then reduces the opinions to reach a consensus on the views (Yousuf, 2007). The researcher

sent out three rounds of questionnaires electronically, and the survey collected the responses anonymously. After the first round, the answers were aggregated and the controlled feedback was shared with the panel of experts for subsequent rounds. Sekayi and Kennedy (2017) expressed that the Likert-type questionnaire provided controlled feedback rather than having the panel communicate with one another, which eliminates groupthink. This classical Delphi study was used to gain insight into future K-12 education PD systems and structure trends.

An essential feature of the Delphi method's multiple rounds of questioning allows the expert panel members to review the collective list of responses to the questionnaire. Then, each expert panel member will rate or evaluate the list of answers based on a predetermined criterion of importance. Finally, the third questionnaire will include a list and the ratings indicated. This would also be indicated if the experts arrived at a consensus during the third round. One benefit is that the experts can revise their opinions or explain their reasoning without arriving at a consensus (Yousuf, 2007). Another benefit of a Delphi study is that the panel of experts remains anonymous. Anonymity reduces the impact of peer pressure to conform and allows all opinions to be considered. Also, according to Fischer (1978), the results gathered from a Delphi study can be used for planning by administrators.

Population

A population is a group that "conforms to specific criteria" (McMillan & Schumacher, 2010, p. 129) to which research results can be generalized. The population for this study is expert PD administrators/curriculum and instruction administrators in California public school districts. Each district's organizational chart differs; however,

this population comprises administrators from any district-level department designing and delivering PD. The CDE (2019) reported 1,037 public school districts.

Sampling Frame

According to Taherdoost (2016), "A sampling frame is a list of the actual cases from which the sample will be drawn" (p. 20). The sampling frame for a research study is the collective group for which the study's data and findings can be generalized. The sampling frame refers to the group of individuals the researcher collected data from for this study (McMillan & Schumacher, 2010). The sampling frame for this study was expert K-12 district-level administrators representing California public school districts in California. The expert K-12 educational administrators who were purposefully selected have worked in the field for at least three years and have been involved with certificated PD.

According to the CDE (2019), 1,037 public school districts provided professional learning for certificated staff. The sampling frame was K-12 district administrators representing all public schools in California. The expert K-12 educational administrators had worked in the field for at least three years and have been involved with certificated PD.

Sample

The population sample is the participants from whom the data is collected (McMillian & Schumacher, 2010). For example, the potential sample population for this study was composed of 15 district administrators working in California public school districts, who, as administrators, successfully lead the implementation of PD in the district. The sample selection process included a purposive method. McMillian

and Schumacher (2010) describe purposive sampling as selecting specific characteristics from the population that will inform the researcher. Delphi studies require a panel of experts who will make up the sample (Dalkey & Helmer, 1963; Yousuf, 2007). The experts chosen for this study were district administrators who have successfully implemented certificated PD in their respective school districts. The criterion used for the selection of these experts were:

- Must currently be a K-12 public education administrator at the district level.
- Must have a minimum of three years of experience as a district administrator.
- Must oversee the implementation of certificated PD.

The sample for this study was 15 expert district administrators from California school districts that have successfully implemented PD for certificated staff.

Sample Selection Process

- Potential participants were identified by utilizing school district websites, listing employees in the curriculum and instruction departments.
- Potential participants who met the criteria were contacted via email (see Appendix B).
- Those who responded were sent the demographic questionnaire via email (see Appendix C).
- 4. Of the individuals who met the criteria, 17 were selected to participate in the study.

- 5. Those who responded and met the demographic criteria were sent the informed consent material (see Appendix D and E).
- 6. Surveys were administered.

Instrumentation

This research study utilized the online survey tool, Survey Monkey, and email to collect data and communicate with the expert panel. The researcher used three rounds of questioning to answer the study's research questions. The first round included an openended question. The second round used the results from round one to create a Likert scale survey that allowed the experts to rate the importance of the round one responses and rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond. The final round allowed the experts to refine their responses and deliver feedback on PD implementation practices.

The researcher developed the surveys using Survey Monkey and emailed the hyperlinked survey to the panel, along with instructions on how to complete the survey.

Round 1

The survey instrument that was used in Round 1 asked the following open-ended question: What do experts in professional development predict the delivery and content of K-12 Education professional development in 2026 and beyond? The researcher coded the responses to the first question and placed the answers into a list to be used in Round 2 (see Appendix F).

Round 2

Next, the expert panel used a six-point Likert scale to rate the importance of the indicated structures of PD delivery and content revealed during Round 1 (see Appendix

G). The ranges on the Likert scale will be: *Very Important, Important, Slightly Important, Slightly Unimportant, Unimportant, and Very Unimportant.* The expert panel was also asked to rate the likelihood of predictions being enacted for each of the indicated structures of PD delivery and content revealed during Round 1. Again, the ranges on the Likert scale will be: *Very Likely, Likely, Slightly Likely, Slightly Unlikely, Unlikely, and Very Unlikely.*

Round 3

The survey instrument that was used in Round 3 contained an open-ended question for each of the highest rated structures of PD delivery and content revealed during Round 2 (see Appendix H). The question was: *How do experts in professional development respond to the predictions made from Research Question 3 and what final suggestions do they make for implementation in 2026 and beyond?*

Validity

According to McMillan and Schumacher (2010), validity determines the extent to which data is credible and trustworthy. Validity also determines whether the research truly measures what it intended to (Bashir et al., 2008). Qualitative researchers have a responsibility to determine the validity of their study by implementing verification strategies during the research. A pilot test in the same format as described in the research design of this study was administered to four district administrators to ensure that the instrument used in this study accurately measured what was intended. These administrators met the same criteria as the sampling for this study. The feedback from the field test volunteers was used to ensure the validity of the surveys. Furthermore, validity

in a Delphi study is assured by the expert consensus achieved after three rounds of surveys and controlled feedback (Yousuf, 2007).

Reliability

Bashir et al. (2008) contend the most important issue in qualitative research is to ensure reliability and validity. Reliability in qualitative research refers to the stability and consistency of measures between responses and multiple coders (McMillan & Schumacher, 2010). A reliable research study's results can be reproduced under the same conditions (Golafshani, 2003; McMillan & Schumacher, 2010). By leveraging the expert panel, this study should produce results that would remain comparable or consistent if this study was used to replicate the findings.

Data Collection

Permission to conduct this research study was granted by the University of Massachusetts Global Institutional Review Board (IRB) (see Appendix I), after the researcher completed the required coursework and was granted a certification from the National Institutes of Health (see Appendix J). Upon receiving IRB consent to collect data, the researcher contacted potential expert panelists by email. The email explained the research purpose to the potential expert panelist, and they were asked to volunteer for the three-round study.

This research study consisted of three questionnaires that were developed to have California K-12 expert district-level administrators describe the predicted structures for PD delivery and content in 2026 and beyond. In addition, three rounds of anonymous surveying took place using Survey Monkey, an online survey program.

Round 1

An email was sent to the expert panel of California K-12 district administrators, outlining an anticipated timeline, a link to the Round 1 survey, and the researcher's contact information. The expert panel was asked to respond to the following open-ended question: "What do experts in professional development predict the delivery and content of K-12 education professional development will look like in 2026 and beyond?" The anonymous responses from Round 1 were compiled into one list of themes and used to prepare the Round 2 survey.

Round 2

Round 2 provided another survey, which was developed from the responses in Round 1. The researcher emailed the expert panel of California K-12 district administrators an anticipated timeline for completion and a link to the Round 2 survey. The expert panel was asked to use a six-point Likert scale to rate the importance of the indicated structures of PD delivery and content revealed during Round 1. The expert panel was also asked to rate the likelihood of predictions being enacted for each of the indicated structures of PD delivery and content revealed during Round 1.

Once the Round 2 surveys were completed, the researcher tallied the score and calculated the mean average for each indicated structure. Next, the researcher organized the themes for each indicated structure of PD delivery and content. Structures were then organized from highest mean score to lowest mean score. The top identified structures in Round 2 were used for Round 3.

Round 3

The researcher identified the top strategies with the highest mean score in Round 2 to develop the Round 3 survey. An email was sent to the expert panel of California K-12 district administrators, including the anticipated timeline for completion, and a link to the Round 3 survey. The expert panel was asked to respond to the following question:

"How do experts in professional development respond to the predictions made from RQ3 and what final suggestions do they make for implementation in 2026 and beyond?"

Upon completion of the Round 3 surveys, the researcher compiled the responses.

Then, the researcher coded and sorted the responses into themes.

Data Analysis

The researcher collected and analyzed data in this Delphi study in three stages. The qualitative data collected from each round was used to develop the questionnaire for the subsequent round. After Round 2, mean scores for each structure were calculated, and the structures were ranked from the highest to lowest mean score. After Round 3, the researcher compiled the responses, coded and sorted the responses into themes. A summary was prepared to describe the top predicted structures for PD delivery and content in K-12 education.

Round 1

Round 1 used Survey Monkey to collect an anonymous response to the following question: "What do experts in professional development predict the delivery and content of K-12 education professional development will look like in 2026 and beyond?" The anonymous responses from Round 1 were compiled into one list of themes and used to prepare the Round 2 survey.

Round 2

Round 2 used Survey Monkey to collect anonymous responses using a six-point Likert scale to rate the importance of the indicated structures of PD delivery and content revealed during Round 1. The expert panel was also asked to rate the likelihood of enacted predictions for each of the indicated structures of PD delivery and content revealed during Round 1. Once the Round 2 surveys were completed, the researcher tallied the score and calculated the mean average for each indicated structure. Next, the researcher organized the themes for each suggested structure of PD delivery, and content and structures were then organized from highest mean score to lowest mean score. The top identified structures in Round 2 will be used for Round 3.

Round 3

The researcher identified the top strategies with the highest mean score in Round 2 to develop the Round 3 survey. The expert panel was asked to respond to the following question: "How do experts in professional development respond to the predictions made from Research Question 3 and what final suggestions do they make for implementation in 2026 and beyond?"

Upon completion of the Round 3 surveys, the researcher compiled the responses. Then, the researcher coded and sorted the responses into themes. The researcher prepared a summary to describe the top predicted structures for PD delivery and content in K-12 education.

Limitations

There were several limitations to this classical Delphi study. First, determining the qualifications of an expert is subjective, and the Delphi methodology requires an

expert panel. Second, the study focused on PD administrators, which could have varying definitions. Additionally, the researcher works in PD and may exhibit bias. Furthermore, this study was limited to California public school districts and did not include charter or private schools. Lastly, the instrumentation used was field-tested but may lack contextual measures.

Summary

Chapter III included an overview, review of the purpose statement, research questions, and research design. The methodology used for the study was a classical Delphi method. The classical Delphi method is a forecasting process framework that uses various rounds of questionnaires sent to a panel of experts (Dalkey & Helmer, 1963; Sitlington & Coetzer, 2015). Next, a description of the population, target population, and sample used for this study were presented. Additionally, information about the instruments, including the field test, validity, and reliability, is presented, followed by a description of data collection, analysis, and limitations. The objective of Chapter III was to describe the rationale for conducting a qualitative research study using a classical Delphi method as the data collection technique.

CHAPTER IV: RESEARCH, DATA COLLECTION, AND FINDINGS

Chapter IV presents the data collected for this study, followed by data analysis.

This study was meant to determine what experts in PD predict delivery and content of K12 education will look like in 2026 and beyond, to rate the likelihood of the predictions
being enacted by 2026 and beyond, and to identify how the experts predict K-12
educational organizations need to structure themselves to deliver predicted PD in 2026
and beyond. Furthermore, Chapter IV reiterates the purpose and research questions for
this study, along with the methodology, population, sample, and presentation of the data.

This chapter will close with a detailed report of the findings of the research study.

Overview

Chapter IV presents the data collected during the Delphi study's various rounds, accompanied by analysis. This study was meant to determine what the experts in PD predict the delivery and content of K-12 education and identify how K-12 educational organizations need to structure themselves so that other districts can build, improve, or implement PD models within their districts. Chapter IV restates the study's purpose and research questions along with the chosen methodology, population, and specific sample before presenting data. Finally, Chapter IV concludes with a summary of the findings.

Purpose Statement

The purpose of this futures Delphi study was to determine what experts in professional development predict the delivery and content of K-12 education will look like in 2026 and beyond, to determine how the experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and

beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted PD in 2026 and beyond.

Research Questions

The following questions were investigated to address the purpose of the study:

- 1. What do experts in professional development predict the delivery and content of K-12 education professional development will look like in 2026 and beyond?
- 2. How do experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond?
- 3. How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for delivery and content of professional development?
- 4. How do experts in professional development respond to the predictions made from Research Question 3, and what final suggestions do they make for implementation in 2026 and beyond?

Research Methods and Data Collection Procedures

This study used the classical Delphi method to collect data from K-12 administrators regarding the delivery and content of PD in K-12 educational organizations. Additionally, the Delphi method allowed the researcher to determine how K-12 educational organizations in California will need to structure themselves to deliver predicted PD in 2026 and beyond. The RAND Corporation developed the Delphi method in 1950 (Dalkey & Helmer, 1963). The classical Delphi method is a forecasting process framework that uses various rounds of questionnaires sent to a panel of experts (Dalkey

& Helmer, 1963; Sitlington & Coetzer, 2015). Consequently, the researcher obtained descriptive data through the questionnaires using the classical Delphi method as a qualitative research design.

This study used the classical Delphi method to gather perceptual data from an expert panel of K-12 administrators who decide on PD structures in their districts. To reach a consensus on the views, this methodology systematically collected information from a group of experts and then reduced the opinions to reach a consensus (Yousuf, 2007). The researcher sent out three rounds of questionnaires electronically, and the survey collected the responses anonymously. After the first round, the answers were aggregated and the controlled feedback was shared with the panel of experts for subsequent rounds. The Likert-type questionnaire provided controlled feedback rather than having the panel communicate with one another, which eliminates groupthink (Sekayi & Kennedy, 2017). Ultimately, this classical Delphi study was used to gain insight into future K-12 education PD systems and structure trends.

An important feature of the Delphi method's multiple rounds of questioning allows the expert panel members to examine the collective list of responses to the questionnaire. Then, each expert panel member will rate or evaluate the list of answers based on a predetermined criterion of importance. Finally, the third questionnaire will include a list and the ratings indicated. This would also be indicated if the experts reached a consensus during the third round. According to Yousuf (2007), one advantage is that the experts can revise their opinions or explain their reasoning without arriving at a consensus. An additional benefit of a Delphi study is that the panel of experts remains anonymous. Anonymity reduces the impact of peer pressure to conform and allows all

opinions to be considered. Finally, according to Fischer (1978), the results gathered from a Delphi study can be used for planning by administrators.

Population and Sample

A population is a group that "conforms to specific criteria" (McMillan & Schumacher, 2010, p. 129) to which research results can be generalized. The population for this study is expert PD administrators/curriculum and instruction administrators in California public school districts. Typically, each district's organizational chart differs; however, this population comprises administrators from any district-level department designing and delivering PD. The CDE (n.d.) reported 1,037 public school districts.

The population sample is the participants from whom the data is collected (McMillian & Schumacher, 2010). As such, the potential sample population for this study would be composed of 15 district administrators working in California public school districts, who, as administrators, successfully lead the implementation of PD in the district. The sample selection process will include a purposive method. Purposive sampling is selecting specific characteristics from the population that will inform the researcher (McMillian & Schumacher, 2010). Delphi studies require a panel of experts who will make up the sample (Dalkey & Helmer, 1963; Yousuf, 2007). The experts selected for this study were district administrators who have effectively implemented certificated PD in their school districts. The criterion used for selecting these experts were:

- Must currently be a K-12 public education administrator at the district level.
- Must have a minimum of three years of experience as a district administrator.
- Must oversee the implementation of certificated PD.

The sample for this study was 15 expert district administrators from California school districts that have successfully implemented PD for certificated staff.

Presentation of the Data

This section presents the data collected for each research question and analysis.

Tables have been embedded to help display the data. The data is presented sequentially, as outlined by the research methodology.

Research Question 1

Research Question 1 asked: What do experts in professional development predict the delivery and content of K-12 education professional development will look like in 2026 and beyond?

Round 1

The researcher began the study by creating an electronic survey using Survey Monkey, which asked the following open-ended question: What do experts in professional development predict the delivery and content of K-12 Education professional development will look like in 2026 and beyond? This initial round was intended to produce an extensive list of possible professional development attributes for delivery and content. The survey was sent out to 17 participants.

Fourteen expert PD administrators responded to this question. After the responses were established, the researcher examined the responses and organized them into a table to begin the coding process and determine themes. Most of the 14 respondents provided multiple qualifications; however, one participant shared only one qualification; the remaining responses varied from four to 14. Some of the responses were simple such as "virtual" or "hybrid," while others were more detailed, like, "In person will involve more

things such as instructional rounds, more modeling, and more in-class support." The researcher coded the data to establish themes before developing the survey for Round 2.

Analysis of Round 1. Fourteen out of the 17 expert PD administrators participated during Round 1 of the study. Again, the data was analyzed and coded which allowed the researcher to identify emerging themes. The research question was split into two categories: predictions for PD delivery and predictions for PD content. The researcher identified 10 different PD delivery themes and 16 PD content areas that emerged from the experts' responses.

Seven of the 14 respondents indicated that PD delivery will be in-person learning opportunities. Further review of the data indicated that six respondents indicated PD delivery would be delivered in a variety of modalities, which could include:

- In-person
- Virtual
- Asynchronous
- Synchronous

Five of the 14 expert PD administrators recognized virtual PD as an acceptable delivery model. Virtual PD can occur via platforms such as Zoom or Google Meet. The PD can be recorded so that participants can refer back to the learning.

Five of the 14 expert PD administrators who responded to the Round 1 questions identified PD would be delivered in a hybrid or blended model. The PD delivery associated with this type of model could be:

- Asynchronous reading and videos.
- Online discussion boards.

- Synchronous class time for collaborative work and discussion.
- Small group and one-on-one time to individualize instruction during in-person sessions.

The participant's list of predictions for PD delivery is outlined in Table 1.

Table 1

List of Possible Professional Development Attributes for Delivery Models

-	Possible Professional Development Attributes	Frequency
1	In-person delivery of professional development	7
2	Virtual delivery of professional development	5
3	Blended (synchronous & asynchronous)	5
4	Self-paced/asynchronous	4
5	Professional development will be data-driven	2
6	Integrated Model	1
7	Personalized professional development	1
8	Job embedded	1
9	Coaching/modeling	1
10	Focus on best practices for instruction	1

Three of the 14 participants stated that PD content will be integrated, rather than PD that focuses on only one content area. Further analysis of the data showed that the integrated content could include:

- Social Emotional Learning (SEL).
- Technology.
- Subject-matter content.
- Pedagogy.

Content specific conceptual knowledge was identified by three of the 14 respondents. Content-specific knowledge refers to subject areas, such as English

language arts, mathematics, science, or social studies. Also, content knowledge generally refers to the facts, concepts, and theories that are taught in academic courses.

Three of the 14 respondents believed that student data analysis would be important content that should be addressed in professional learning opportunities. PD focused on student data analysis could include:

- Predictive analytics for student success.
- Disaggregating student data into trackable skills and goals.
- Using short cycle assessments to inform instruction.
- Processes for data analysis and determining next steps.

Two of the four respondents said PD needs to emphasize skills. An emphasis on skills requires teachers to learn and refine the pedagogies necessary to teach students these skills. Students need skills such as communication, collaboration, critical thinking, and creativity to thrive in the 21st century.

Two of the 14 respondents identified the importance of differentiated content in PD for educators. Blended PD would be an example of a method to provide differentiated PD opportunities effectively. The list for PD content areas is outlined in Table 2. Items with the same frequency were added at random.

Table 2
List of Possible Professional Development Content Areas

-	Possible Professional Development attributes	Frequency
1	Content-specific conceptual knowledge	3
2	Data analysis	3
3	Emphasis on skill	2
4	Digital integration	2
5	Social Emotional Learning (SEL)	2
6	Digital Literacy	2
7	Driven by economic climate	2
8	Differentiated content	2
9	Instructional strategies	2
10	Standards-based grading	1
11	Project-based learning	1
12	Calibration of grading practices	1
13	Digital integration	1
14	Universal Design for Learning (UDL)	1
15	Social Emotional Learning (SEL)	1
16	Digital Citizenship	1
17	Driven by district initiatives and priorities	1
18	Driven by social climate	1
19	Ethnic studies	1

Research Question 2

Research Question 2 asked: How do experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond?

Round 2

During this round, the researcher prepared a second electronic survey through Survey Monkey in an effort to allow the 14 expert PD administrators to rate the likelihood of enacting the 10 different PD delivery themes and 16 PD content areas that emerged from the experts' responses. The survey contained the following prompt: *Using the five-point Likert scale provided below, please rate the likelihood of the aggregate responses obtained from Round 1 surrounding the predicted professional development delivery modalities and content in K-12 education.*

As stated previously, all 17 expert PD administrators were invited to participate during this round. Of the 17, there were 14 participating. The researcher's primary goal for this round was to determine the top five PD modalities for delivery and the top five content areas in an effort to gather data for the third and final round. After presenting the ten PD delivery themes and 16 PD content areas, the researcher asked participants to rank each one on a five-point Likert scale. The data from this round were then placed into two tables and a mean was calculated for each (see Table 3 and Table 4).

Table 3

Mean Ranking of Professional Development Attributes for Delivery Models

Professional Development Delivery Models	Mean
Blended (synchronous & asynchronous)	4.85
Professional development will be data-driven	4.64
Professional development will focus on best practices for instruction	4.61
Virtual delivery of professional development	4.57
Self-paced/asynchronous	4.35
In-person delivery of Professional development	4.07
Integrated Model	4.07
Personalized professional	4
Job embedded	3.64
Coaching/modeling	3.57

Table 4Mean Ranking of Professional Development Content Areas

Professional Development Content	Mean
Driven by district initiatives and priorities	4.42
Data analysis	4.28
Content-specific conceptual knowledge	4.23
Differentiated content	4.21
Universal Design for Learning (UDL)	4.21
Standards-based grading	4.21
Social Emotional Learning (SEL)	4.15
Digital Literacy	4.15
Digital integration	4.14
Ethnic studies	4.14
Calibration of grading practices	4.07
Project-based learning	4
Emphasis on skill	3.92
Digital Citizenship	3.92
Driven by social climate	3.64
Driven by economic climate	3

Analysis of round 2. Fourteen of the 17 participants involved with the study participated in Round 2. Six of the 26 questions were answered by only 13 participants. The participants were asked to rank the predictions of all enacting the 10 different PD delivery themes and 16 PD content areas that emerged from the experts' responses during Round 1 on a five-point Likert scale ranging from *Not Likely at All* to *Very Likely*. Each point on the Likert scale was then given a point value ranging from 1 point for *Not Likely at All* up to 5 points for *Very Likely*. The mean was able to be established by assigning a point value to each point on the Likert scale. The mean scores for PD delivery models ranged from 4.85-3.57 and the mean scores for content ranged from 4.42-3.

PD administrators predicted that the highest quality for PD, with a mean score of 4.85 is:

• Blended (asynchronous and blended)

Second to that, the data shows that with a mean score of 4.64, the second highest predicted quality of PD is:

• Data-driven

Next, the data indicated with a mean score of 4.61, a model for PD will:

• Focus on best practices for instruction

The data showed with a mean score of 4.35, PD will be:

• Self-paced/asynchronous

Finally, the data identified with a mean score of 4.07 two predictions:

- Integrated model
- In-person delivery

Table 5 shows the highest rated PD delivery predictions.

Table 5Highest Rated Professional Development Delivery

Professional Development Delivery Models	Mean
Blended (synchronous & asynchronous)	4.85
Professional development will be data-driven	4.64
Professional development will focus on best practices for instruction	4.61
Self-paced/asynchronous	4.35
In-person delivery of professional development	4.07
Integrated Model	4.07

The PD content area highest rated by PD administrators, with a mean score of 4.42 will be:

• Driven by district initiatives and priorities

The second highest rated PD content area, with a mean score of 4.28 will be:

• Data analysis

Next, with a mean of 4.23, PD administrators indicated:

• Content-specific conceptual knowledge

Finally, PD administrators identified three content areas with a mean score of 4.21:

- Differentiated content
- Universal Design for Learning (UDL)
- Standards-based grading

Table 6 shows the highest rated PD content predictions.

Table 6Highest Rated Professional Development Content Predictions

Professional Development Content	Mean
Driven by district initiatives and priorities	4.42
Data analysis	4.28
Content-specific conceptual knowledge	4.23
Differentiated content	4.21
Universal Design for Learning (UDL)	4.21
Standards-based grading	4.21

The lowest rated of the predictions for K-12 PD delivery, with a mean score of 3.57 was:

Coaching/modeling

The second lowest rated predicted delivery model, with a mean score of 3.64 was:

• Job-embedded

The predicted least likely PD content quality, with a mean score of 3 included:

• PD driven by the economic climate

Finally, with a mean of 3.64, the next lowest rated PD content quality was:

• PD driven by the social climate

Table 7 shows the lowest ranked predicted PD delivery and content.

 Table 7

 Lowest Ranked Professional Development Delivery and Content

Professional Development Content	Mean
Job embedded	3.64
Coaching/modeling	3.57
Driven by social climate	3.64
Driven by economic climate	3

The purpose of Round 2 during this Delphi study was to predict the top five qualities for K-12 PD delivery and content from the aggregated data from Round 1 and utilize that information in creating the Round 3 survey question in efforts to answer the research questions associated with this study.

As previously stated, the goal of Round 2 was to narrow the list of 26 predictions for K-12 PD to the top five for delivery and content. The top five predictions for delivery and content were returned to the panel of expert PD administrators in order to petition responses to answer the final research questions in this study (see Table 8).

Table 8Top Five K-12 Professional Development Predictions for Professional Development Delivery and Content

Professional Development	Rank
Delivery Method	
Blended (synchronous & asynchronous)	1
Professional development will be data-driven	2
Professional development will focus on best practices for instruction	3
Self-paced/asynchronous	4
Integrated Professional development model	5
<u>Content</u>	
Driven by district initiatives and priorities	1
Data analysis	2
Content-specific conceptual knowledge	3
Differentiated content	4
Universal Design for Learning (UDL)	5

Research Question 3

Research Question 3 asked: How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for delivery and content of professional development?

Round 3

The final round of this study took the five top-rated predictions from the previous round and provided the expert panel the opportunity to answer three open-ended questions. The first question was: *How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for the delivery of professional development?* The top five predictions for PD delivery: Blended (asynchronous & synchronous), Data-driven, Focus on best

practices for instruction, Self-paced asynchronous learning, and Delivery will be an integrated model, addressing multiple topics in one PD.

The second question was: How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for the content of professional development? The top five predictions for PD content were: Driven by district initiatives and priorities, Data analysis, Emphasis on skill, Differentiated content, and Universal design of learning.

The final question was: How do experts in professional development respond to the predictions made from Research Question 3 and what final suggestions do they make for implementation in 2026 and beyond? The top five predictions for PD delivery were: Blended (asynchronous & synchronous), Data-driven, Focus on best practices for instruction, Self-paced asynchronous learning, and Delivery will be an integrated model, addressing multiple topics in one PD. The top five predictions for PD content were: Driven by district initiatives and priorities, Data analysis, Emphasis on the skill, Differentiated content, and Universal design of learning. The panel was asked to describe what they believe will be important for school districts to successfully implement these predictions in the content and delivery of PD in their districts.

Analysis of Round 3. All 17 expert PD administrators were invited once again to participate during this round. Of the 17, there were nine respondents. The data from Round 3 was qualitatively analyzed, coded, and arranged into a frequency table in an effort to identify the themes that a district may implement in order to change its PD content and structures based on the predictions.

In analyzing the data for how districts need to structure themselves in order to deliver Blended (synchronous/asynchronous) PD, the most frequent concept that surfaced from the expert panel, with a frequency of 2, was that of:

Provide after-school learning opportunities, including short PDs after school
or longer PD on Saturdays. After-school PD is convenient for teachers
because they do not need to make lesson plans for substitute teachers.
 Teachers who need more time for learning would be able to attend the longer
learning sessions offered on Saturdays.

With a frequency of 2, the PD administrators reported the theme of:

 Design learning opportunities that are short and concise. Providing PD after school would allow teachers to learn in short periods of time and the information can be condensed to be more concise. These PD opportunities are ideal for veteran teachers who have already experienced many cycles of initiatives in education and only need the essential information to add to their robust pedagogical toolkits.

The least recurring themes, at a frequency of 1, were:

- Blended PD should be inclusive.
- Blended PD should occur over 2 days.

Table 9 displays the structures suggested by the panel of experts that may be implemented when designing blended (synchronous/asynchronous) PD learning opportunities.

Table 9Top Structures for Implementing K-12 Professional Development Predictions for Blended Professional Development Delivery

Structures for Blended Professional Development Delivery Models	Frequency
After school learning opportunities	2
Short and concise after school professional development	2
Inclusive	1
Occur over two days	1

Prediction 2 was identified during Round 2, and its structures for implementation were discovered during Round 3 are presented below.

With Prediction 2, PD delivery models will be *Data-driven*, the panel of experts produced a theme that recurred three times for a response percentage of 33%. The prediction was:

 Use student data from standardized assessments to decide the target areas of student need. This data will determine the PD offerings.

The next most significant themes had a frequency rate of two and was mentioned 22% of the time. The themes were:

- Qualitative data should be gathered during instructional round walkthroughs within the district.
- District leaders should observe other high performing districts.

The last theme had a frequency rate of one, and a response rate of 11%. The last theme was:

 Research based practices must be applied to effective PD. District leaders should choose the best practices that have the highest effect size on increasing student achievement. Table 10 displays the structures suggested by the panel of experts that may be implemented when using data to determine PD delivery models.

Table 10Top Structures for Implementing K-12 Professional Development Predictions for Datadriven Professional Development Delivery

Structures for Data-Driven Professional Development Delivery Models	Frequency
Student need	3
Instructional round walkthroughs	2
Observe high performing districts	2
Research-based best practices for student achievement	1

During Round 2, the expert panel found *Focus on Best Practices for Instruction* to be a predictor of what will drive PD delivery models. During Round 3, the expert panel identified three themes for integrating best practices into PD delivery models. Each theme had a frequency of 1 or an 11% frequency rate. These three themes were:

- Develop units of instruction for teachers that include research-based best practices.
- Determine what content can be delivered asynchronously and which content must be taught synchronously.
- Backwards map PD and apply best practices.

Table 11 displays strategies suggested by the panel of experts that may be applied when designing the delivery model of PD that focuses on best practices for instruction.

Table 11

Top Structures for Implementing K-12 Professional Development Predictions for Professional Development that focuses on Best Practices for Instruction

Structures for Professional Development Delivery Models Focused on Best Practices	Frequency
Develop units of study	1
Determine which content can be delivered synchronous versus asynchronous	1
Backwards map and apply best practices	1

Again, during Round 2, the respondents made predictions that would help other district administrators lead PD implementation. During that round, that same panel overwhelmingly provided responses and identified one particular theme over and over again with a frequency of 6 or 66% of responses; that theme was:

 Provided asynchronous, on-demand learning so that teachers can learn based on their self-identified needs.

There were only two other themes that surfaced. The next most significant theme had a frequency rate of 2 and was mentioned 22% of the time. The theme was:

 Asynchronous learning can be job-embedded. Teachers would be able to access the learning during their prep period, or before and after instruction.

The last theme had a frequency rate of one, and a response rate of 11%. The last theme was:

Districts need to use a Learning Management System (LMS) to organize the
content and collect data. The LMS allows teachers to easily navigate the
content. Administrators will be able to access data, such as course completion
rates and accuracy.

Table 12 displays structures suggested by the panel of experts that may be implemented when designing self-paced learning opportunities for teachers.

Table 12Top Structures for Implementing K-12 Professional Development Predictions for Self-paced, Asynchronous Learning

Structures for Self-paced, Asynchronous Professional Development	
Delivery	Frequency
Provide on-demand, asynchronous professional development to meet	
self-identified needs	6
Asynchronous professional development can be job-embedded	2
Find a Learning Management System to support the needs	1

The last structure identified during Round 2 was an *Integrated Professional*Development Model. During Round 3, the expert panel spoke about two themes, yielding a frequency of two and a response rate of 22%:

- Hire staff developers and teachers on special assignment (TOSA) to focus on creating PD that addresses multiple content areas or district initiatives.
- Provide opportunities for staff developers and TOSAs to collaborate across
 instructional departments. This collaboration will be intentional and will allow
 for the creation of integrated content in one PD session.

Lastly, the respondents identified one other theme, with a frequency of 1 the theme was:

 Provided access to differentiated content in PD. PD that has a variety of content areas, an integrated approach, should also be differentiated for teachers. Table 13 displays structures suggested by the panel of experts that may be implemented when designing PD with content that is integrated, addressing multiple content areas, for teachers.

Table 13Top Structures for Implementing K-12 Professional Development Predictions for a Model Focused on Integrated Content

Structures for Integrated Professional Development Delivery Models	Frequency
Hire staff developers and Teachers on Special Assignment	2
Collaborate across instructional departments	2
Provide differentiated content	1

The next data set analyzed the data for how districts need to structure themselves in order to implement the top rated predictions for PD content. With a frequency of 3, PD that has *Content Driven by District Initiatives and Priorities* should:

 Focus on content alignment to district initiatives and priorities. The alignment should be present and explicitly called out in each PD opportunity.

With a frequency of 2, the PD administrators reported the theme of:

Communication with the school leaders is essential. District priorities should
be revisited at leadership meetings throughout the year. There should be
constant reminders of where the district is headed and the content of PD will
support this journey.

The least recurring themes, at a frequency of 1, were:

 Content driven by district initiatives and priorities should be determined through strategic planning. Districts need to prioritize initiatives and be aware of the initiative overload teachers may feel. Prioritizing initiatives will allow district leaders to make informed decisions when planning PD content.

Table 14 displays the structures suggested by the panel of experts that may be implemented when designing PD driven by district initiatives and priorities.

Table 14Top Structures for Implementing Professional Development Content Driven by District Initiatives and Priorities

Structures for Implementing PD Content Driven by District Initiatives and Priorities	Frequency
Content alignment to district initiatives and priorities	3
Communication with school leaders	2
Strategic planning	1
Prioritize district initiatives and understand initiative overload	1

Prediction 2 was identified during Round 2, and its structures for implementation were discovered during Round 3 are presented below.

With Prediction 2, PD content will be driven by *Data Analysis*, the panel of experts produced a theme that recurred three times for a response percentage of 33%. The prediction was:

The needs for PD would be determined based on the data analysis. PD
offerings would only be available if the data demonstrated a need in this
content area.

The next most significant themes had a frequency rate of two and was mentioned 22% of the time. The themes were:

- Data must be progress monitored in order for PD content to reflect the current needs.
- District leadership must engage in conversations with the site leadership. The
 data will guide the topics of the conversations and in turn the PD offerings to
 teachers.
- Share the data with all stakeholders. It will be important for teachers to connect the PD to how it will improve student achievement in areas needing growth.

Table 15 displays the structures suggested by the panel of experts that may be implemented when using data analysis to drive PD content.

Table 15Top Structures for Implementing K-12 Professional Development Content Driven by Data Analysis

Structures for Implementing Professional Development Content Driven	
by Data Analysis	Frequency
Determine the needs for professional development offerings based on data analysis	3
Data analysis through progress monitoring	2
Data driven conversations with leadership	2
Communicate what the data is showing to all stakeholders	2

During Round 2, the expert panel found *Emphasis on Skill* to be a predictor of what will drive PD content. During Round 3, the expert panel identified the top theme for PD content. The theme had a frequency of 3 or a 33% frequency rate. This theme was:

 Collect PD feedback from sites, reflect on the data, and pivot or continue, based on findings. With a frequency of 2, the PD administrators reported the theme of:

Implement universal screener to determine the skills that need development.
 Provide PD based on identified needs.

The least recurring themes, at a frequency of 1, were:

- Teachers need skills in data literacy, in order to determine the content students need to develop skills.
- There must be collaboration across instructional departments when creating
 PD content that supports skills.

Table 16 displays strategies suggested by the panel of experts that may be applied when designing the PD content that emphasizes instruction based on skill.

Table 16

Top Structures for Implementing K-12 Professional Development Predictions for Professional Development content with an Emphasis on Skill

Structures for Implementing Professional Development Content with an Emphasis on Skill	Frequency
Collect professional development feedback from sites and pivot as needed	3
Implement universal screeners to determine the skills that need development	2.
Teachers need skills in data literacy	1
Cross-departmental collaboration on professional development content	1
development development	1

Again, during Round 2, the respondents made predictions that would help other district administrators lead PD implementation. During that round, that same panel overwhelmingly provided responses and identified each theme with a frequency of 1 or 11% of responses; those themes were:

 Provide teachers with opportunities for differentiated learning through model lessons and classroom observations.

- Identify direct connections between different content areas and the
 differentiated PD opportunities that would support the learning. Recognize
 that not all PD should be one-size-fits all. Teachers in different grade levels,
 content areas, and even school sites have unique needs.
- Provide clarity around what learning success looks like in the classroom. This
 will need to be differentiated for our different learners in one classroom,
 including English language learners, special education, and even foster and
 homeless students.
- After each PD provides a follow-up email with takeaways and addresses
 questions that came up during that particular session.

Table 17 displays structures suggested by the panel of experts that may be implemented when designing PD with differentiated content.

Table 17Top Structures for Implementing Differentiated Content in K-12 Professional Development

Structures for Implementing Differentiated Content	Frequency
Opportunities for differentiation through model lessons and classroom observations	1
Identify direct connections between different classrooms and different professional development offerings	1
Provide clarity around what learning success in classroom looks like	1
After each professional development provide a follow-up email with take aways and address questions that came up during the professional development	1

The last structure identified during Round 2 was implementing *Content that is Universally Designed for all Learners*. During Round 3, the expert panel spoke about two themes, yielding a frequency of one and a response rate of 11%:

- Inclusivity should be present in all content. Every learner should be represented and have access to the standards. PD should support these goals.
- Student social-emotional learning should be addressed in PD. Teachers need access to information and strategies to support the well-being of their students.

Table 18 displays structures suggested by the panel of experts that may be implemented when designing PD content that is universally designed for all learners.

Table 18Top Structures for Implementing K-12 Professional Development Content that is Universally Designed

Structures for Implementing Differentiated Content that is Universally	
Designed	Frequency
Inclusivity should be present in all content	1
Student social-emotional learning should be addressed in professional \	
development content	1

Research Question 4

Research Question 4 asked: How do experts in professional development respond to the predictions made from Research Question 3 and what final suggestions do they make for implementation in 2026 and beyond?

Research Question 4 Analysis

The last question was an open-ended question that solicited final thoughts from the expert panel. The instrument asked: "How do experts in professional development respond to the predictions made from Research Question 3 and what final suggestions do they make for implementation in 2026 and beyond?" Nine expert PD administrators responded to the open-ended questions.

In analyzing the data from the responses to the predictions and suggestions made, the most frequent concept that surfaced from the expert panel, with a frequency of 3, was that of:

 The predictions from the expert panel are relevant to the needs in districts regarding K-12 PD.

With a frequency of 2, the experts reported the themes of:

- Acknowledge the gaps in the data and respond to the identified needs. Provide teachers with learning opportunities to support student achievement in these areas.
- Apply research-based practices to teaching adult learners. Engage adults by using andragogy strategies to teach adult learners.
- Collaborate with the teachers' union. Negotiate required PD that is built into the teachers' contract.
- Districts need to be consistent with implementing PD opportunities.
- Instruction in the classroom looks different now and so should PD. Teachers
 are no longer solely providing direct instruction and they are required to
 facilitate. Strategies used during PD should model this.
- PD should be offered outside of instructional time so that teachers can stay in the classroom. The more instructional time the teacher misses, the more chances students have for missing learning opportunities.
- Offer on-demand, asynchronous PD for all teachers. This allows teachers to learn based on self-identified needs.

- Analyze student data and create PD offerings based on the findings. PD should be data-driven and progress monitored.
- The COVID-19 pandemic has created unique challenges for administrators
 and teachers. PD offerings should consider these challenges and respond. For
 example, teachers need more PD on social-emotional learning and traumainformed practices because of the trauma and loss students experienced during
 the pandemic.

The least recurring themes, at a frequency of 1, were:

- Alignment with district goals should be present in all PD.
- Be mindful of initiative overload and prioritize the PD offerings, ensuring they support district priorities.
- Emphasis on skill is important and PD should integrate strategies that support teachers with skills that can be applied across all content areas.
- Feedback from stakeholders is critical and it should drive the decision-making surrounding PD.
- Provide opportunities for teachers so that they can observe best practices and learn from other teachers on their campus by allowing them to participate in model lessons.

Table 19 displays each of the identified themes and their frequency count.

Table 19 *Expert Responses to Predictions*

Expert Responses to Predictions	Frequency
Predictions are relevant	3
Acknowledge the gaps in the data and respond	2
Apply research-based practices to teaching adult learners	2
Collaborate with the teachers union	2
Districts need to be consistent with implementation	2
Instruction in the classroom looks different now, so professional development should look different	2
Keep teachers in classrooms and offer professional development outside of instructional hours	2
On-demand, asynchronous professional development should be available	2
Professional development should be based on the needs of the students	2
Response to gaps in learning from the pandemic will pose ongoing challenges	2
Alignment with district goals	1
Be mindful of initiative overload	1
Emphasis on skill is important	1
Feedback from stakeholders is critical	1
Model lessons for teachers	1

Summary

Chapter IV offered the data and an analysis of this classical Delphi study. The purpose of this futures Delphi study was to determine what experts in PD predict the delivery and content of K-12 education will look like in 2026 and beyond, to determine how the experts in PD rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted PD in 2026 and beyond.

The subsequent research questions directed the study:

- 1. What do experts in professional development predict the delivery and content of K-12 education professional development will look like in 2026 and beyond?
- 2. How do experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond?
- 3. How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for delivery and content of professional development?
- 4. How do experts in professional development respond to the predictions made from Research Question 3, and what final suggestions do they make for implementation in 2026 and beyond?

Round 1 of the student was a qualitative round designed to petition as many examples of possible PD attributes for delivery and content. The sample size consisted of 17 expert district administrators; 14 of these experts responded to the question sent to them via electronic survey in Survey Monkey during Round 1. The information collected during Round 1 answered Research Question 1. The information was then structured into a chart, analyzed, and coded and then utilized to create the survey question for Round 2.

Round 2 of this Delphi study was also quantitative as the participants were asked to rank the aggregate responses from Round 1 on a five-point Likert scale ranging from *Not Likely at All* to *Very Likely*. The research question was split into two categories: predictions for PD delivery and predictions for PD content. The researcher identified 10 different PD delivery themes and 16 PD content areas that emerged from 14 experts'

responses. The mean for each ranking was then calculated and placed in a chart. Finally, the researcher sorted the mean for each prediction from high to low, finding the top five predictions for PD delivery and content and answering Research Question 2.

Round 3 was also a quantitative round meant to generate expert opinions from the participants on how to best structure PD to enact the predictions for PD content and delivery. The answers of nine respondents were organized into charts, analyzed, and coded. Then, the researchers placed the data into frequency charts for each prediction for PD delivery and content. In all, 17 themes for PD delivery and 18 themes for PD content were identified, again, separated into ten charts. The data collected during this round would serve K-12 school districts looking to improve their current structures for PD delivery to students and/or content.

Round 3 asked a final open-ended question, which allowed the experts to provide final thoughts and suggestions to the predictions made in Round 1 and Round 2 surveys. The answers of nine respondents were organized into charts, analyzed, coded, and placed into frequency charts. 15 themes emerged and 10 of the themes reiterated themes from previous rounds. This round answered Research Questions 3 and 4.

The purpose of Chapter IV was to present information surrounding the data related to this Delphi study on the content and delivery of PD in K-12 school districts. To finalize this study, Chapter V will present conclusions, implications, and recommendations for future research.

CHAPTER V: FINDINGS, CONCLUSIONS, AND RECOMMENDATIONS

Chapter V provides a review of this classical Delphi study's purpose statement, research questions, methodology, and the study's population and sample. Furthermore, this chapter provides the researcher's findings, conclusions, implications for action, and recommendations for future research. The chapter concludes with the researcher's final remarks and reflections on the research study.

Purpose Statement

The purpose of this futures Delphi study was to determine what experts in professional development predict the delivery and content of K-12 education will look like in 2026 and beyond, to determine how the experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted professional development in 2026 and beyond.

Research Questions

The study was guided by the following research questions:

- 1. What do experts in professional development predict the delivery and content of K-12 education professional development will look like in 2026 and beyond?
- 2. How do experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond?
- 3. How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for delivery and content of professional development?

4. How do experts in professional development respond to the predictions made from Research Question 3 and what final suggestions do they make for implementation in 2026 and beyond?

Methodology

The methodology chosen for this study was the classical Delphi method. The study collected data from K-12 administrators regarding the delivery and content of PD in K-12 educational organizations. The Delphi methodology allowed the researcher to determine how K-12 educational organizations in California will need to structure themselves to deliver predicted PD in 2026 and beyond. Round 1 was to allow the panel of experts to make predictions about the content and the delivery of PD. After analysis, their responses were used to create the Round 2 survey instrument. The purpose of Round 2 was to rank, on a five-point Likert scale, the major themes discovered during Round 1. The experts were to identify the top five predictions for both, PD delivery and content.

There were 17 expert PD administrators identified to participate in the study. During Round 1, a total of 14 expert PD administrators responded (82%). The research question was split into two categories: Predictions for PD delivery and predictions for PD content. After qualitatively analyzing and coding the data, the researcher identified 10 different PD delivery themes and 16 PD content areas that emerged from the experts' responses. During Round 2, a total of 14 administrators responded (82%). Their responses allowed the researcher to determine the mean of each of the 26 previously identified predictions from Round 1. That information was used to identify the top five predictions for PD delivery and the top five predictions for PD content. Round 2 data allowed the researcher to create the Round 3 survey instrument. During Round 3, a total

of nine PD administrators responded to the survey (52%). In all, 17 themes for PD delivery and 18 themes for PD content were identified, again, separated into ten charts. Round 3 also asked a final open-ended question, which allowed the experts to provide final thoughts and suggestions to the predictions made in Round 1 and Round 2 surveys. The answers of nine respondents were organized into charts, analyzed, coded, and placed into frequency charts. Fifteen themes emerged and 10 of the themes reiterated themes from previous rounds.

Major Findings

This section of Chapter V presents the major findings of the study. These findings will be presented for the research questions identified in this chapter.

Research Questions

Research Question 1

The first research question associated with Delphi study was: What do experts in professional development predict the delivery and content of K-12 Education professional development will look like in 2026 and beyond?

Round 1. The major finding associated with this initial round was the 10 different PD delivery themes and 16 PD content areas identified by the expert panel. The top 10 PD delivery themes, ranked in order from the most frequent to the least frequent are:

- 1. Blended (synchronous & asynchronous).
- 2. Data-driven.
- 3. Focused on best practices for instruction.
- 4. Virtual delivery of PD.
- 5. Self-paced/asynchronous.

- 6. In-person.7. Integrated.
- 8. Personalized.
- 9. Job embedded.
- 10. Coaching/modeling.

The top 16 PD content themes, ranked in order from the most frequent to the least frequent are:

- 1. Driven by district initiatives and priorities.
- 2. Data analysis.
- 3. Content-specific conceptual knowledge.
- 4. Differentiated content.
- 5. Universal Design for Learning (UDL).
- 6. Standards-based grading.
- 7. Social Emotional Learning (SEL).
- 8. Digital literacy.
- 9. Digital integration.
- 10. Ethnic studies.
- 11. Calibration of grading practices.
- 12. Project-based learning.
- 13. Emphasis on skill.
- 14. Digital citizenship.
- 15. Driven by social climate.
- 16. Driven by economic climate.

Research Question 2

The second research question of this classical Delphi study was: *How do experts* in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond?

Round 2. The major finding associated with this round was the mean calculation of each identified of the 26 qualifications from Round 1. The researcher then sorted the scores from high to low, subsequently identifying the top five predictions for professional development delivery and content. The top five predictions for PD delivery, ranked in order are:

- 1. Blended (synchronous and asynchronous).
- 2. Data-driven.
- 3. Focus on best practices.
- 4. Virtual PD.
- 5. Self-paced/asynchronous.

The top five predictions for PD content, ranked in order are:

- 1. Driven by district initiatives and priorities.
- 2. Data analysis.
- 3. Content-specific conceptual knowledge.
- 4. Differentiated content.
- 5. Universal Design for Learning (UDL).

Research Question 3

The third research question for this Classical Delphi student was: *How do experts* in professional development predict K-12 school districts will need to structure their

organizations to implement the top five rated predictions for delivery and content of professional development?

Research Question 4

The fourth, and final research question for this study was: *How do experts in professional development respond to the predictions made from Research Question 3, and what final suggestions do they make for implementation in 2026 and beyond?*

Round 3. The major finding to Research Question 3 was the collective predictions of the expert panel on the content and delivery of PD structures identified during Round 2 to help other districts provide exemplary PD in their districts. There were 17 themes for PD delivery, and 18 themes for PD content were identified, again, separated into 10 charts.

Prediction 1: Professional Development Delivery Models will Deliver Content

Through a Blended (synchronous/asynchronous) Model. There were four
recommendations from the expert panel that support the PD delivery model that will
focus on blended learning opportunities. Ranked order from most frequently suggested to
least frequently suggested, the four expert recommendations are:

- Provide after-school learning opportunities, including short PDs after school
 or longer PD on Saturdays. After-school PD is convenient for teachers
 because they do not need to make lesson plans for substitute teachers.
 Teachers who need more time for learning would be able to attend the longer
 learning sessions offered on Saturdays.
- 2. Design learning opportunities that are short and concise. Providing PD after school would allow teachers to learn in short periods, and the information can

be condensed to be more concise. These PD opportunities are ideal for veteran teachers who have already experienced many cycles of initiatives in education and only need the essential information to add to their robust pedagogical toolkits.

- 3. Blended PD should be inclusive.
- 4. Blended PD should occur over two days.

Prediction 2: Professional Development Delivery Models will be Data-Driven.

There were four recommendations from the expert panel that support the PD delivery model that will focus on best practices for instruction. Ranked order from most frequently suggested to least frequently suggested, the four expert recommendations are:

- Use student data from standardized assessments to decide the target areas of student need. This data will determine the PD offerings.
- 2. Qualitative data should be gathered during instructional round walkthroughs within the district.
- 3. District leaders should observe other high performing districts.
- Research based practices must be applied to effective PD. District leaders
 should choose the best practices that have the highest effect size on increasing
 student achievement.

Prediction 3: Professional Development Delivery Models will Focus on Best

Practices for Instruction. There were three recommendations from the expert panel that support the PD delivery model that will focus on best practices for instruction.
Ranked order from most frequently suggested to least frequently suggested, the three expert recommendations are:

- Develop units of instruction for teachers that include research-based best practices.
- 2. Determine what content can be delivered asynchronously and which content must be taught synchronously.
- 3. Backwards map PD and apply best practices.

Prediction 4: Professional Development Delivery Models will Provide Self-

Paced, Asynchronous Learning. There were three recommendations from the expert panel that support the PD delivery model that will provide self-paced, asynchronous learning. Ranked order from most frequently suggested to least frequently suggested, the three expert recommendations are:

- Provided asynchronous, on-demand learning so that teachers can learn based on their self-identified needs.
- 2. Asynchronous learning can be job-embedded. Teachers would be able to access the learning during their prep period, or before and after instruction.
- 3. Districts need to use a learning management system to organize the content and collect data. The learning management system allows teachers to easily navigate the content. Administrators will be able to access data, such as course completion rates and accuracy.

Prediction 5: Professional Development Delivery Models will Deliver

Professional Development with Integrated Content. There were three recommendations from the expert panel that support the PD delivery model that will provide PD with integrated content. Ranked order from most frequently suggested to least frequently suggested, the three expert recommendations are:

- Hire staff developers and teachers on special assignment to focus on creating
 PD that addresses multiple content areas or district initiatives.
- Provide opportunities for staff developers and teachers on special assignment to collaborate across instructional departments. This collaboration will be intentional and will allow for the creation of integrated content in one PD session.
- Provided access to differentiated content in PD. PD that has a variety of content areas, and an integrated approach, should also be differentiated for teachers.

The expert panel made four predictions for PD content. The predictions and recommendations are:

Prediction 6: Professional Development Delivery Models will Have Content

That is Driven by District Initiatives and Priorities. There were four

recommendations from the expert panel that support the PD content that is driven by

district initiatives and priorities. Listed in rank order from most frequently suggested to

least frequently, the four expert recommendations are:

- 1. Focus on content alignment to district initiatives and priorities. The alignment should be present and explicitly called out in each PD opportunity.
- Communication with the school leaders is essential. District priorities should
 be revisited at leadership meetings throughout the year. There should be
 constant reminders of where the district is headed and the content of PD will
 support this journey.

- 3. Content driven by district initiatives and priorities should be determined through strategic planning.
- 4. Districts need to prioritize initiatives and be aware of the initiative overload teachers may feel. Prioritizing initiatives will allow district leaders to make informed decisions when planning PD content.

Prediction 7: Professional Development Delivery Models that will Determine

Content After Engaging in Data Analysis. There were four recommendations

from the expert panel that support the PD content that is determined after data analysis.

Listed in rank order from most frequently suggested to least frequently, the four expert recommendations are:

- The needs for PD would be determined based on the data analysis. PD
 offerings would only be available if the data demonstrated a need in this
 content area.
- 2. Data must be progress monitored in order for PD content to reflect the current needs.
- 3. District leadership must engage in conversations with the site leadership. The data will guide the topics of the conversations and in turn the PD offerings to teachers.
- 4. Share the data with all stakeholders. It will be important for teachers to connect the PD to how it will improve student achievement in areas needing growth.

Prediction 8: Professional Development Content will Emphasize Skills. There were four recommendations from the expert panel that support PD content that emphasizes skills. Listed in rank order from most frequently suggested to least frequently, the four expert recommendations are:

- Collect PD feedback from sites, reflect on the data, and pivot or continue, based on findings.
- Implement a universal screener to determine the skills that need development.
 Provide PD based on identified needs.
- Teachers need skills in data literacy in order to determine the content students need to develop skills.
- There must be collaboration across instructional departments when creating PD content that supports skills.

Prediction 9: Professional Development Content will be Differentiated for all

Learners. There were four recommendations from the expert panel that support differentiated PD content. Listed in rank order from most frequently suggested to least frequently, the four expert recommendations are:

- Provide teachers with opportunities for differentiated learning through model lessons and classroom observations.
- 2. Identify direct connections between different content areas and the differentiated PD opportunities that would support the learning. Recognize that not all PD should be one-size-fits all. Teachers in different grade levels, content areas, and even school sites have unique needs.

- Provide clarity around what learning success looks like in the classroom. This
 will need to be differentiated for our different learners in one classroom,
 including English language learners, special education, and even foster and
 homeless students.
- 4. After each PD provides a follow-up email with takeaways and addresses questions that came up during that particular session.

For All Learners. There were two recommendations from the expert panel that support PD content that is universally designed for all learners. Listed in rank order from most frequently suggested to least frequently, the two expert recommendations are:

- Inclusivity should be present in all content. Every learner should be represented and have access to the standards. PD should support these goals.
- 2. Student social-emotional learning should be addressed in PD. Teachers need access to information and strategies to support the well-being of their students.

The major finding with respect to Research Question 4 was the collective responses and suggestions to the identified predictions. There were 15 collective themes that emerged and 10 of the themes reiterated themes from previous rounds. Ranked order from most frequently suggested to least frequently suggested, the responses are:

- The predictions from the expert panel are relevant to the needs in districts regarding K-12 PD.
- Acknowledge the gaps in the data and respond to the identified needs. Provide teachers with learning opportunities to support student achievement in these areas.

- 3. Apply research-based practices to teaching adult learners. Engage adults by using andragogy strategies to teach adult learners.
- 4. Collaborate with the teachers' union. Negotiate required PD that is built into the teachers' contract.
- 5. Districts need to be consistent with implementing PD opportunities.
- 6. Instruction in the classroom looks different now and so should PD. Teachers are no longer solely providing direct instruction and they are required to facilitate. Strategies used during PD should model this.
- 7. PD should be offered outside of instructional time so that teachers can stay in the classroom. The more instructional time the teacher misses, the more chances students have for missing learning opportunities.
- 8. Offer on-demand, asynchronous PD for all teachers. This allows teachers to learn based on self-identified needs.
- Analyze student data and create PD offerings based on the findings. PD should be data-driven and progress monitored.
- 10. The COVID-19 pandemic has created unique challenges for administrators and teachers. PD offerings should consider these challenges and respond. For example, teachers need more PD on social-emotional learning and trauma-informed practices because of the trauma and loss students experienced during the COVID-19 pandemic.
- 11. Alignment with district goals should be present in all PD.
- 12. Be mindful of initiative overload and prioritize the PD offerings, ensuring they support district priorities.

- 13. Emphasis on skill is important and PD should integrate strategies that support teachers with skills that can be applied across all content areas.
- 14. Feedback from stakeholders is critical and it should drive the decision-making surrounding PD.
- 15. Provide opportunities for teachers so that they can observe best practices and learn from other teachers on their campus by allowing them to participate in model lessons.

Unexpected Findings

There were two unexpected findings associated with this classical Delphi study. First, the comparison of the most frequently mentioned PD delivery predictions in Round 1 to the ratings of the most important of the delivery predictions in Round 2 became an unexpected finding. According to the frequency of which they were mentioned, the highest rated predictions for PD delivery were identified in Round 1 based on frequency scores (f) were:

- In-person delivery of PD (f = 7).
- Virtual delivery of PD (f = 5).
- Blended (synchronous & asynchronous) (f = 5).
- Self-paced/asynchronous (f = 4).
- PD will be data-driven (f = 2).
- Integrated Model (f =1).
- Personalized PD (f =1).
- Job embedded (f =1).

- Coaching/modeling (f =1).
- Focus on best practices for instruction (f = 1).

Only three of the most frequently identified predictions from the above list - *Blended (synchronous & asynchronous), Virtual deliver of PD and Self-paced/asynchronous* - were rated amongst the list of the most likely to be implemented into PD delivery models. This unexpected finding suggests that the factors that were most prevalent in Round 1 were not the only predictions for PD delivery models that were the most important because other predictions ranked higher in Round 2.

Second, two of the identified factors for PD content in Round 2 should all be integrated into all PD within districts. The two content areas - *Differentiated content and Universal design for learning (UDL)* - are elements of PD that will provide access to the content for all participants, no matter their experience or education levels. This finding was surprising because although these are strategies teachers use in the classroom daily, they are typically not strategies applied to PD.

Conclusions

The purpose of this futures Delphi study was to determine what experts in PD predict the delivery and content of K-12 education will look like in 2026 and beyond, to determine how the experts in PD rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted PD in 2026 and beyond.

Conclusion 1

After reviewing the study's findings, the first conclusion is the importance of providing differentiated learning opportunities for all teachers. PD content will need to include the knowledge and skills teachers must have to increase students' achievement. The content must meet teachers' diverse needs and support developing teachers' areas of interest to continue developing life-long learners. Furthermore, the PD delivery methods must vary to meet diverse learners' needs. Too often, district-provided PD is a one-size-fits-all experience. In order to meet the needs of all teachers, PD offered by the school district should be provided synchronously, asynchronously, and blended. Therefore, it is concluded that, in the future, PD must be tailored to the individual teachers' specific needs.

Conclusion 2

PD content and delivery should be driven by student data. The expert panel was clear, explaining student data analysis should be multi-faceted. The different measures will include predictive analytics, short-cycle assessments, and disaggregated data. As such, all of the student data will inform the PD opportunities offered to teachers. Therefore, it is concluded that, in the future, PD must be driven by specific identified needs based upon student data.

Conclusion 3

The final conclusion of this study is that when district leaders are designing PD opportunities, they should focus on best practices that have the highest effect size on increasing student achievement based on John Hattie's research. PD aims to build teacher efficacy, which is the teacher's belief in their ability to bring about desired outcomes in

student achievement. PD that will build teacher efficacy and increase teacher effectiveness will focus on best practices such as aligning instruction to the, standards, designing formative assessments, and providing meaningful student feedback. Therefore, it is concluded that, in the future, PD must be designed using proven models, such as Hattie's, that have a history of producing the highest effect on student achievement.

Implications for Action

Teacher PD plays an integral role in increasing student achievement. Multiple legislation acts, including ESEA and NCLB raised the standards and accountability, which caused administrators to analyze the PD offered to better meet the needs of educators and students. The rigor of the Common Core State Standards accentuates the call for teachers to be highly skilled and impact student outcomes. As such, this section provides two implications for action for districts to consider.

Implications for Action 1

The study focused on determining what experts in PD predict the delivery and content of K-12 education will look like in 2026 and beyond and the likelihood of successful implementation. The findings and conclusions of this study indicated that there are many attributes that already exist in the delivery and content districts offered for PD. Many districts currently offer PD on content-specific knowledge or standards-based grading; however, typically, these offerings are one-size-fits-all. Therefore, districts must design differentiated professional learning opportunities and creativity leverage technology to help support successful implementation. The experts in the study indicated that blended learning, which includes synchronous and asynchronous learning opportunities, would be an emerging delivery method that allows easier access to

differentiated content. That being said, districts must explore learning management systems that could provide a platform that is easy to navigate for the teacher and student. Districts are missing a significant opportunity to impact desired learning outcomes for teachers.

Implications for Action 2

The second implication for action to be presented in this section is that a district needs to establish processes and structures to collect and analyze data pertaining to PD. Not only should student data drive decision-making, PD feedback data should also be considered when designing the delivery and content of learning opportunities. Districts need to create one standard PD feedback form that is aligned to district priorities and require the feedback form is completed after each PD session. Also, districts need to establish short-term assessment cycles for analyzing the PD feedback data. PD facilitators should make changes to content and delivery based on the data. Finally, districts should look for trends for improvement in student achievement that correlates to the target PD offered. Continual data monitoring will determine what PD is needed and how the PD will support student outcomes, thus improving the overall effectiveness of district PD provided.

Implications for Action 3

The final implication for action to be presented in this section is that each district must have one person responsible for PD to ensure focus and direction for PD are consistent. Very few districts in California have one central person responsible for all PD in the district. Typically, districts have several people responsible for different content areas of PD, which can allow for inconsistency and create barriers to providing high-

quality PD to all teachers. As such, employing a Director of Professional Development would allow for the implementation of a cohesive PD plan, which would increase the quality of the PD provided for teachers.

Recommendations for Further Research

The following areas for further research denote the findings and limitations associated with this classical Delphi study on PD. K-12 PD exists in every school district and can benefit from additional research. The following list represents recommendations for future research:

- Replicate this study in different school districts from different counties or states. This study was limited to 10 school districts: Apple Valley, Alvord, Corona-Norco, Hemet, Lawndale, Moreno-Valley, Orange, Palo-Alto, Riverside, and Upland, in California.
- Recreate this Delphi study but focus on the staff development specialists, who generally design and deliver PD to teachers.
- Recreate this study but use site administrators rather than district administrators.
- Replicate this Delphi study with only representation from academically highperforming districts.
- A similar study that identifies any differences between elementary and secondary content and delivery structures.
- A study that analyzes district office structures for implementation of PD.
- Conduct a study to determine whether there would be a difference in findings from classroom teachers.

- Conduct a study with educational consultants who provide PD for school districts.
- Conduct a study to determine whether blended or asynchronous professional learning has a greater impact than traditional in-person PD on student achievement.

Concluding Remarks and Reflections

Educational reform has consistently acknowledged the need for ongoing professional learning, but the policies have never specifically identified how to provide effective PD. As such, each school district in California determines how to structure PD opportunities differently, spending the funds in various ways, which is a difficult task for any district. As far as I could determine, districts have an idea of some of the elements that contribute to high-quality PD, but they do not strategically implement these structures for PD content and delivery. I contemplate if these predictions were more open-minded to innovative delivery methods, such as blended learning, virtual or asynchronous learning, because of the COVID-19 pandemic and the need to dramatically shift teaching and learning. This study has opened my eyes to the slow evolution of PD and revealed there is still so much more work to be done.

The major findings of this study will allow for districts to create strategic plans for providing high-quality PD to all teachers. Meeting the needs of all teachers is a difficult task with many different challenges that arise along the way. The predictions for the content and delivery of PD found over the course of this study are meant to help district leaders create a formal plan for providing differentiated, accessible PD that helps teachers provide high-quality instruction where all students reach their academic goals.

REFERENCES

- Al-Shammari, Z., Faulkner, P. E., & Forlin, C. (2019). Theories-based inclusive education practice. *Education Quarterly Reviews*, 2(2), 408-414.
- Anderson, S. E. (2006). The school district's role in educational change. *International journal of educational reform*, 15(1), 13-37.
- Archambault, L., Wetzel, K., Foulger, T. S., & Kim Williams, M. (2010). Professional development 2.0: Transforming teacher education pedagogy with 21st century tools. *Journal of Digital Learning in Teacher Education*, 27(1), 4-11.
- Bashir, M., Afzal, M. T., & Azeem, M. (2008). Reliability and validity of qualitative and operational research paradigm. *Pakistan journal of statistics and operation* research, 35-45.
- Bastin, L. L. (2003). Professional development and teacher learning styles: Selected

 Kentucky teachers' perceptions about school-based or state -designed

 professional development (Publication No. 3074489) [Doctoral Dissertation,

 University of Kentucky]. ProQuest Dissertations & Theses Global.
- Beach, R. (2012). Can online learning communities foster professional development? Language Arts, 89(4), 256-262.
- Blackburn, S. (2020). Taking learning to new heights: Poway USD's blended learning and professional development environment increases student engagement and the use of education technology. https://www.districtadministration.com/taking-learning-to-new-heights/

- Blank, R. K., de las Alas, N., & Council of Chief State School, O. (2009). The effects of teacher professional development on gains in student achievement: How meta-analysis provides scientific evidence useful to education leaders (ED544700).

 ERIC. https://files.eric.ed.gov/fulltext/ED544700.pdf
- California Department of Education. (n.d.). *Professional learning. Professional learning Quality schooling framework*. Retrieved August 21, 2021. https://www.cde.ca.gov/qs/pl/
- Carbaugh, B. G., Marzano, R. J., Toth, M. D., Houpt, K., & Sahadeo-Turner, T. (2015).

 School leadership for results: Shifting the focus of leader evaluation [Book].

 Learning Sciences International.
- Chism, R. L. (2000). A Vygotskian perspective on electronic bulletin boards: An exploratory study (Publication No. 9971740) [Doctoral Dissertation]. ProQuest Dissertations & Theses Global.
- Commission on Teacher Credentialing. (2009). California standards for the teaching profession. https://www.ctc.ca.gov/docs/default-source/educator-prep/standards/cstp-2009.pdf
- Corcoran, T. B. (1995). Helping teachers teach well: Transforming professional development. https://doi.org/10.1037/e383832004-001
- Dalkey, N., & Helmer, O. (1963). An experimental application of the Delphi method to the use if experts. *Management Science*, *9*(3), 458-467.
- Darling-Hammond, L., & McLaughlin, M. W. (1995). Policies that support professional development in an era of reform [Article]. *Phi Delta Kappan*, 76(8), 597-604.

- Dede, C., & Harvard University, G. S. of E. (2006). *Online professional development for teachers: Emerging models and methods*. Harvard Education Press.
- Disch, S. E. H. (2020). *Teachers' perceptions of the contribution of core features of professional development to learning and teaching practice* [Doctoral Dissertation]. ProQuest Dissertations and Theses Global.
- DuFour, R. (2004). What is a "professional learning community"? *Educational leadership*, 61(8), 6-11.
- DuFour, R. & Eaker, R., 2009. Professional learning communities at work tm: best practices for enhancing students achievement. Solution Tree Press.
- EdSource, I. P. A. C. A. (1998). Shifting the focus to learning: California's accountability debates. State experts discuss how accountability can improve student achievement. *EdSource Report*.
- Elmore, R. F. (2004). *School reform from the inside out: Policy, practice, and performance*. Harvard Education Press.
- Farr, W. J., & Saltmarsh, S. (2018). Priming your environment for growth through a collaborative model of instructional coaching. *Educational Renaissance*, 7, 23-28.
- Fischer, R. G. (1978). The Delphi method: A description, review, and criticism. *Journal of Academic Librarianship*, 4(2), 64–70.
- Foster, M. (2004, January). An innovative professional development program for urban teachers. *Phi Delta Kappan*, 40, 1-6.

- Francois van, A. (2017). Communities of practice as a tool for continuing professional development of technology teachers' professional knowledge. *International Journal of Technology and Design Education*, 28(2), 417-430.
- Frerichs, S. W., Fenton, M. S. P., & Wingert, K. (2018). A model for out-of-school educator professional learning [article]. *Adult Learning*, 29(3), 115-122.
- Fullan, M. (2010). All systems go: The change imperative for whole system reform.

 Corwin Press.
- Fullan, M. (2016). The elusive nature of whole system improvement in education. *Journal of Educational Change*, 17(4), 539-544.
- Gissy, C. L. (2010). Elementary teachers' attitudes about professional development:

 Professional development schools versus non-professional development schools.

 [Doctoral Dissertation]. ProQuest Dissertations and Theses Global.
- Golafshani, N. (2003). Understanding reliability and validity in qualitative research. *The Qualitative Report*, 8(4), 597-607.
- Graham, G., Houston Independent School District, D. o. R., & Accountability. (2020).

 Office of Special Education Services (OSES) professional development survey

 results, 2019-2020. Research Educational Program Report.
- Grant, P. A., Young, E. E., Montbriand, C., & North Central Regional Educational Lab,
 O. B. I. L. (2001). *Professional development for teachers of reading* (ED464074).

 ERIC. https://www.govinfo.gov/content/pkg/ERIC-ED464074/pdf/ERIC-ED464074.pdf

- Grosz, R. T. (2004). Research -supported professional development: Perceptions of K–12 educators of self -determined professional development needs (Publication No. 3154777) [Doctoral Dissertation]. ProQuest Dissertations & Theses Global. Ann Arbor.
- Guskey. T. R. (1994). *Professional development in education: In search of the optimal mix* (ED369181). ERIC. https://files.eric.ed.gov/fulltext/ED369181.pdf
- Guskey, T. R. (2002). Professional development and teacher change. *Teachers and Teaching*, 8(3), 381-391.
- Guskey, T. R. (2009). Closing the knowledge gap on effective professional development. *Educational Horizons*, 87(4), 224-233.
- Guskey, T. R., & Yoon, K. S. (2009). What works in professional development., *Kappan Journal*.
- Hargreaves, A. (1995). *Development and desire: A postmodern perspective* (ED372057). ERIC. https://files.eric.ed.gov/fulltext/ED372057.pdf
- Hattie, J. (2012). Visible learning for teachers: Maximizing impact on learning.

 Routledge.
- Haug, B. S., & Mork, S. M. (2021, 04 April). Taking 21st century skills from vision to classroom: What teachers highlight as supportive professional development in the light of new demands from educational reforms. *Teaching and Teacher Education*, 100.
- Hightower, A. M. (2002). District bureaucracy supports culture of learning. *Center for the Study of Teaching and Policy, University of Washington*.

- Hirsh, S. (2007). NSDC standards and tools help STRENGTHEN professional development. *Developing a Staff of Learners*, 19(1).
- Holme, R. (2019). An investigation into teacher-initiated or diy professional development: The push and pull of teacher professional development. https://discovery.dundee.ac.uk/en/studentTheses/an-investigation-into-teacher-initiated-or-diy-professional-devel
- Hunzicker, J. (2011). Effective professional development for teachers: A checklist.

 *Professional Development in Education, 37(2), 177-179.
- Jimenez, L. (2020). Preparing American students for the workforce of the future:

 Ensuring Every student's readiness for college, career, and civic life. *Center for American progress*.
- John, M. (2014). Designing and implementing effective professional learning.

 Corwin.
- Klein, J. I., Rice, C. (2014). *U.S. education reform and national security*. Ukraine: Council on Foreign Relations.
- Knight, J. I. M. (2019/). Students on the margins: How instructional coaching can increase engagement and achievement [Article]. *Learning Professional*, 40(6), 28-32.
- Knowles, M. S. (1984). The adult learner: A neglected species. Gulf.
- Knowles, M. S., Holton, E. F., & Swanson, R. A. (2005). The adult learner: The definitive classic in adult education and human resource development. Elsevier.

- Krutka, D., Carpenter, J., & Trust, T. (2017). Enriching professional learning networks:

 A framework for identification, reflection, and intention [Article]. *TechTrends:*Linking Research & Practice to Improve Learning, 61(3), 246-252.
- Latz, A. O., Neumeister, K. L. S., Adams, C. M., & Pierce, R. L. (2009). Peer coaching to improve classroom differentiation: Perspectives from project clue. *Roeper Review*, 31(1), 27-39.
- Les, F. (2013). Peer coaching: Unlocking the power of collaboration [Book]. Corwin.
- Lin, P. (2013). Studying teachers' degree of classroom implementation, teachers'

 Implementation practices, and students' learning as outcomes of K-12 STEM professional development (Publication No. 3587839) [Doctoral Dissertation].

 ProQuest Dissertations & Theses Global.
- Luke, A., & McArdle, F. (2009). A model for research-based State professional development policy [Article]. *Asia-Pacific Journal of Teacher Education*, *37*(3), 231-251.
- Lutrick, E. & Szabo, S. (2012). Instructional leaders' beliefs about effective professional development. *Delta Kappa Gamma Bulletin*, 78(3), 6-12.
- McCarthy, J, & Riley, S. (2000). A new vision for teaching professional development. *Leadership*, 30(2), 34-36.
- McKnight, G. L. (2018). Exploring the relationship between professional development leaders' competencies of effective professional learning and teachers' perceptions of Professional Development (Publication No. 10789710) [Doctoral Dissertation]. ProQuest Dissertations & Theses Global.

- McMillian, J., & Schumacher, S., (2010). *Research in education evidence-based inquiry* (7th ed.). Pearson.
- Miles, K. H., & Guiney, E. (2000). School districts' new role [Article]. *Education Week*, 19(40), 30.
- National Commission on Teaching & America's Future (NCTAF). (1996). What matters most: Teaching for America's future. National Commission on Teaching and America's Future. https://www.teachingquality.org/wp-content/uploads/2018/04/What_Matters_Most.pdf
- National Staff Development Council. (2010). NSDC's definition of professional development [Website]. Retrieved https://sedl.org/pubs/sedl-letter/v19n01/nsdc-standards-tools.html.
- Neil, R. (1986). Eleven traditional methods of inservice teacher education. https://doi.org/10.1080/01626620.1985.10519249
- Nishimura, T. (2014). Effective professional development of teachers: A guide to actualizing inclusive schooling. *International Journal of Whole Schooling*, 10(1), 19-42.
- Novota, M. R. C. (2003). The administrator's role in professional development: The connection between the district's strategic plan and professional development for best teaching practices with emphasis at the middle school (Publication Number 3114450). [Ed.D., Northern Illinois University]. ProQuest Dissertations & Theses Global. Ann Arbor

- Okere, D. (2011). *Professional development for k–12 charter school teachers in Jersey City: Effects on student achievement* (Publication No. 3450163) [Doctoral Dissertation]. ProQuest Dissertations & Theses Global.
- Pharis, T. J., Wu, E., Sullivan, S., & Moore, L. (2019). Improving teacher quality:

 Professional development implications from teacher professional growth and
 effectiveness system implementation in rural Kentucky high schools. *Educational Research Quarterly*, 42(3), 29-48.
- Pokhrel, T. R., & Behera, S. K. (2016). Expectations of teachers from teachers professional development program in Nepal. *American Journal of Educational Research*, 4(2), 190-194.
- Potchka, R. (2009). Essential elements of high quality professional development for teachers of two elementary school districts (Publication No. 3355844) [Doctoral Dissertation]. ProQuest Dissertations & Theses Global.
- Reyna, S. D. (2019). A qualitative case study of teachers' perceptions of professional learning communities as effective teacher professional development for teaching and learning (Publication No. 27667363) [Doctoral Dissertation, Grand Canyon University]. ProQuest Dissertations & Theses Global.
- Rodman, A. (2018). Learning together, learning on their own: What if schools could offer teachers both shared professional learning experiences and personalized learning opportunities? *Educational Leadership*, 76(3), 12-18.
- Roth, W.-M., & Lee, Y.-J. (2007). 'Vygotsky's neglected legacy': Cultural-historical activity theory. *Review of Educational Research*, 77(2), 186.

- Sekayi, D., Kennedy, A. (2017, Oct 2017 2018-10-19). Qualitative Delphi method: A four round process with a worked example. *The Qualitative Report*, 22(10), 2755-2763.
- Sharratt, L., & Fullan, M. (Eds.). (2009). Realization: The change imperative for Deepening district-wide reform. Corwin Press.
- Showers, B., & Joyce, B. (1996). The evolution of peer coaching. *Educational Leadership*, 53(6), 12–16.
- Sitlington, H., & Coetzer, A. (2015, 2021-09-09). Using the Delphi technique to support curriculum development. *Education & Training*, *57*(3), 306-321.
- Skinner, B. F. (1968). *The technology of teaching*. Prentice-Hall.
- Smith, S. E. (2016). The use of micro-blogging for teacher professional development support and personalized professional development. ERIC.
- Southerland, S. A., Granger, E. M., Hughes, R., Enderle, P., Ke, F., Roseler, K., ... & Tekkumru-Kisa, M. (2016). Essential aspects of science teacher professional development: Making research participation instructionally effective. *AERA open*, 2(4), 2332858416674200.
- Svendsen, B. (2020). Inquiries into teacher professional development—What matters?. *Education*, *140*(3), 111-130.
- Taherdoost, H. (2016). Sampling methods in research methodology: How to choose a sampling technique for research. *International Journal of Academic Research in Management*, 5(2), 18-27.
- Tate, M. L. (2012). "Sit and get" won't grow dendrites: 20 professional learning strategies that engage the adult brain. Corwin Press.

- Tong, W., & Razniak, A. (2017, 2017 2021-09-11). Building professional capital within a 21st century learning framework. *Journal of Professional Capital and Community*, 2(1), 36-49.
- Tucker, D. L. (2006). *Defining the requisite knowledge for providers of in-service*professional development for K–12 teachers of science: Refining the construct

 (Publication Number 3247636) [Ed.D., University of La Verne]. ProQuest

 Dissertations & Theses Global. Ann Arbor
- Vadeboncoeur, J. A. (2005). *Child development and the purpose of education: A historical context for constructivism in teacher education*. In Constructivist teacher education (pp. 25-48). Routledge.
- U.S. Department of Education. (n.d.). Every Student Succeeds Act (ASSA). https://www.ed.gov/essa?src=ft
- U. S. Department of Education. (2006). Answering the challenge of a changing world: Strengthening education for the 21st Century. https://doi.org/10.1037/e498412006-001
- U.S. Department of Education. (2007). No Child Left Behind Act. https://doi.org/10.1037/e414462005-001
- U.S. Department of Education. (2008). *A nation accountable: Twenty-five years after A Nation at Risk*. https://doi.org/10.1037/e565352006-001
- U.S. Department of Education. (2017). Race to the top fund: Purpose [Website].

 Retrieved July 1, 2022, from https://www2.ed.gov/programs/racetothetop-district/index.html#:~:text=The%20Race%20to%20the%20Top%2DDistrict%20p

- rogram% 20supports% 20bold% 2C% 20locally, students% 20through% 20a% 20pers on alized% 20approach.
- U.S. Department of Education. (2018). School improvement fund: Purpose [Website].

 Retrieved July 1, 2022, from https://www2.ed.gov/programs/sif/index.html
- Vygotsky, L. (1978). Interaction between learning and development. *Readings on the Development of Children*, 23(3), 34-41.
- Whittaker, A., Snyder, J., & Freeman, S. (2001). Restoring balance: A chronology of the development and uses of the California standards for the teaching profession [article]. *Teacher Education Quarterly*, 28(1), 85-107.
- World Economic Forum. (2020). The future of jobs report 2020. Geneva: World Economic Forum.
- Yoon, S. W.-Y., Scarloss, B., & Shapley, K. L. (2007). Reviewing the evidence on how teacher professional development affects student achievement. *Issues & Answers*, 033.
- Yousuf, M. I. (2007). Using experts opinions through Delphi technique. *Practical Assessment, Research, and Evaluation*, 12(1), 4.
- Zambak, V. S., Alston, D. M., Marshall, J. C., & Tyminski, A. M. (2017).
 Convincing science teachers for inquiry-based instruction: Guskey's staff
 development model revisited. *Science Educator*, 25(2), 108-116.

APPENDICES

APPENDIX A

Synthesis Matrix

Source	Methodology	Theoretical Frameworks	Hstorical Perspecitves	The Impact of Federal Reforms on Professional Development Policy	Professional Development Effective Practices	Professional Development Delivery Models	Barriers to Effective Professional Development	21st Century Learning
(Al-Shammari et al., 2019)		X						
(Anderson, 2006)					X			
(Archambault et al., 2010)					X	х		X
(Bashir et al., 2008)	x							
(Bastin, 2003)					X			
(Beach, 2012)					X	X		X
(Blackburn, 2020)					Х	Х		
(Blank et al., 2009)						J.	х	
(Carbaugh et al., 2015)					100			Х
(Chism, 2000)		X			X	х		
(Corcoran, 1995)				X	X	X	Х	X
(CDE, n.d.)		X			X	Х		
(CSTP, 2009)		X			X	15		
(CTC, 2009)		X			X			Х
(Dalkey & Helmer, 1963)	х						V	
(Darling-Hammond & McLaughlin, 1995)					X	x	х	~
(Dede, 2006)					X	X		x
(Degartment of Education, 1983)			X	X		Α		Α
(Department of Education, 1983) (Department of Education, 2006)			X	X				
(Department of Education, 2008)			Λ.	X		X		
(Disch, 2020)		X	X		х	X		
(DuFour, 2004)		Α	Α		X		х	
(DuFour & Eaker, 2009)					X	X		
(Elmore, 2002)					X	X		
(Farr & Saltmarsh, 2018)					X	X		
(Fischer, 1978)	Х					Α		
(Francois van, 2017)					х	X		
(Frerichs et al., 2018)		x				X	X	
(Fullan, 2010)					X	X	A	
	2	_ s	_ s	The Impact of deral Reforms Professional velopment Poli	sional pment Practice	onal nent fodels	Effecti sional pment	Zi B
Source	Methodology	Theoretica	Hstorica	The Impact of Federal Reforms on Professional Development Policy	Profes Develo Effective	Professi Developr Delivery N	Barriers to Effective Professional Development	21st Cent Learnin
Source (Fullan, 2016)		Theoretical	Hstorical Perspecitives	The Im Federal Re Profes Developm	Professional Development Effective Practices	Professional Development Delivery Models	Barriers to X Profes Develo	21st Century Learning
	X Methodolog	Theoretica	Hstorica Perspecity	The Im Federal R Profes Developm	Profes Develo Effective I	Professi Developi Delivery N	x	21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020)								21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001)		Theoretica Framework	X	X	Profes Develo Effective	Professi Developt Delivery N	x	21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004)					X		x	21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994)			X	X	X X		x	21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002)			X	X	X X X	X	x	21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009)			X	X	X X X X	X	x x x	21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grose, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009)			X	X	X X X X	X X X	x	21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995)			X	X	X X X X X X	X	x x x	21st Cent Learnin
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012)			X	X	X X X X	X X X	x x x	
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995)			X	X	X X X X X X	X X X	x x x x	Z1st Cent
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021)			X	X	X X X X X X	X X X X	x x x x	
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Gusky, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002)		X	X	X	X X X X X X X X X	X X X X	x x x x	
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 2004) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2019) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007)		X	X	X	X X X X X X X X X X	X X X X	x x x x	
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Gusky & Yoon, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019)		X	X	X	X X X X X X X X X X	X X X X	X X X X X X	
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2020) (Grant et al., 2020) (Grosz, 2004) (Guskey, 2004) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019) (Hunzicker, 2011) (Jimenez, 2020) (John, 2014)		X	X	X	X X X X X X X X X X	X X X X	X X X X X X	x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Graham et al., 2021) (Grosz, 2004) (Guskey, 2002) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2019) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019) (Hunzicker, 2011) (Jimenez, 2020) (John, 2014) (Klein and Rice, 2014)		X	X	X	X X X X X X X X X X X X X X X X X X X	X X X X X X	X X X X X X	x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019) (Hunzicker, 2011) (Jimenez, 2020) (John, 2014) (Klein and Rice, 2014)		X X X	X	X	X X X X X X X X X X	X X X X	X X X X X X	x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hightower, 2002) (Holme, 2019) (Hunzicker, 2011) (Jimenez, 2020) (John, 2014) (Klein and Rice, 2014) (Knight, 2019) (Knowles et al., 2005)		X	X	X	X X X X X X X X X X X X X X X X X X X	X X X X X X	X X X X X X X	x x x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2021) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattle, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019) (Hunzicker, 2011) (Jimenez, 2020) (John, 2014) (Klein and Rice, 2014) (Klein and Rice, 2014) (Knight, 2019) (Knowles et al., 2005) (Krutka et al., 2017)		X X X	X	X	X X X X X X X X X X X X X X X X	X X X X X X X	X X X X X X	x x x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Graham et al., 2001) (Grosz, 2004) (Guskey, 2002) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019) (Hunzieker, 2011) (Jimenez, 2020) (John, 2014) (Klein and Rice, 2014) (Knight, 2019) (Knowles et al., 2005) (Krutka et al., 20017) (Les, 2013)		X X X	X	XXX	X X X X X X X X X X X X X X X X X X X	X X X X X X	X X X X X X X	x x x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Graham et al., 2001) (Grosz, 2004) (Guskey, 2002) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019) (Hunzicker, 2011) (Jimenez, 2020) (John, 2014) (Klein and Rice, 2014) (Knight, 2019) (Krutka et al., 2005) (Krutka et al., 2017) (Les, 2013) (Lin, 2013)		X X X	X	X	X X X X X X X X X X X X X X X X	X X X X X X X	X X X X X X X	x x x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2001) (Grant et al., 2001) (Grosz, 2004) (Guskey, 2002) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019) (Hunzicker, 2011) (Jimenez, 2020) (John, 2014) (Klein and Rice, 2014) (Knight, 2019) (Knowles et al., 2005) (Krutka et al., 2017) (Les, 2013) (Lin, 2013) (Lin, 2013)		X X X	X	XXX	X X X X X X X X X X X X X X X X X X X	X X X X X X X	X X X X X X X	x x x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hightower, 2002) (Holme, 2019) (Holme, 2019) (Holme, 2019) (Hon, 2014) (Klein and Rice, 2014) (Knight, 2019) (Knowles et al., 2005) (Krutka et al., 2017) (Les, 2013) (Liue & McArdle, 2009) (Lutrick & Szabo, 2012)		X X X	X	XXX	X X X X X X X X X X X X X X X X X X X	X X X X X X X	X X X X X X X X X X	x x x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 2004) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hirsh, 2007) (Holme, 2019) (Hunzieker, 2011) (Jimenez, 2020) (John, 2014) (Klein and Rice, 2014) (Knight, 2017) (Les, 2013) (Lin, 2013) (Lin, 2013) (Luke & McArdle, 2009) (Lutrick & Szabo, 2012) (McCarthy & Riley, 2000)		X X X	X	XXX	X X X X X X X X X X X X X X X X X X X	X X X X X X X	X X X X X X X X X X X X X	x x x
(Fullan, 2016) (Golafshani, 2003) (Graham et al., 2020) (Grant et al., 2001) (Grosz, 2004) (Guskey, 1994) (Guskey, 2002) (Guskey, 2009) (Guskey, 2009) (Hargreaves, 1995) (Hattie, 2012) (Haug & Mork, 2021) (Hightower, 2002) (Hightower, 2002) (Holme, 2019) (Holme, 2019) (Holme, 2019) (Hon, 2014) (Klein and Rice, 2014) (Knight, 2019) (Knowles et al., 2005) (Krutka et al., 2017) (Les, 2013) (Liue & McArdle, 2009) (Lutrick & Szabo, 2012)		X X X	X	XXX	X X X X X X X X X X X X X X X X X X X	X X X X X X X	X X X X X X X X X X	x x x

Source	Methodology	Theoretical Frameworks	Hstorical Perspecitves	The Impact of Federal Reforms on Professional Development Policy	Professional Development Effective Practices	Professional Development Delivery Models	Barriers to Effective Professional Development	21st Century Learning
(Nishimura, 2014)	7325					X		
(NCTAF, 1996)					X	X		
(Neil, 1986)			X			X		
(Novota, 2003)					Х	x		х
(Okere, 2011)				X				
(Pharis et al., 2019)						X		
(Pokhrei & Behera, 2016)					X	X		
(Potchka, 2009)			-		X			
(Reyna, 2019)				X				
(Rodman, 2018)		6			X	X		
(Roth & Lee, 2007)		X						
(Sekayi & Kennedy, 2017)	Х							
(Sharratt & Fullan, 2009)					х			
(Sitlington & Coetzer, 2015)	X							
(Skinner, 1968)		X						
(Smith, 2016)					X			
(Southerland et al., 2016)		X						
(Svendsen, 2020)		X			X	X		
(Taherdoost, 2016)	Х							
(Tate, 2012)						X	X	х
(Tong & Razniak, 2017)		х			X	x	х	х
(Tucker, 2006)					X			V 100 MI
(Vadeboncoeur, 2005)								х
(Vygotsky, 1978)		X						- 11555
(Weidenfeld & Bashevis, 2013)					X	х		
(Whittaker et al.,2001)		х	e e) i			
(World Economic Forum, 2020)			U.					х
(Yoon et al., 2007)					X	X	X	3-7.0
(Youself, 2007)	X							
(Zambak et al., 2017)					X	×		

APPENDIX B

Invitation to Participate

STUDY: A Delphi Study of the delivery and content of K-12 educational professional development and the likelihood of the predictions being enacted by 2026.

Date

Dear Prospective Study Participant,

You are invited to participate in a research study to identify the delivery and content of K-12 educational professional development and the likelihood of the predictions being enacted by 2026. The leading investigator for this study is Christalle Hart, Doctoral Candidate at the University of Massachusetts, Global (UMass Global) Doctor of Education in Organizational Leadership program. You were selected to participate in this study because of your expertise and experience in K-12 education and professional development.

PURPOSE: The purpose of this futures Delphi study is to determine what experts in professional development predict the delivery and content of K-12 education will look like in 2026 and beyond, to determine how the experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted professional development in 2026 and beyond.

PROCEDURES: If you decide to participate in this study, you will receive three rounds of electronic surveys via Survey Monkey, with each survey taking approximately 15 to 20 minutes to complete. The Round 1 survey will contain an open-ended question. The Round 2 survey will utilize a Likert scale survey where participants will rate the recommendations identified from the first-round survey. Finally, the Round 3 survey will contain open-ended questions about the recommendations that were rated highest.

RISKS, INCONVENIENCES, AND DISCOMFORTS: This study involves no more than minimal risk, and no known harms or discomforts are associated with this study. There is no cost to you for participating, and you will not be compensated for your participation. The survey will be completed anonymously, and the researcher will not know your identity.

POTENTIAL BENEFITS: Your participation in this study does not yield any direct benefits to you. However, analysis of the data generated from your participation in this study may contribute to the design and delivery of professional development in K-12 education. The information from this study is intended to inform researchers and educational leaders. Additionally, the findings from this study will be made available to all participants.

ANONYMITY: Records of information that you provide for the research study, and any personal information you provide, will not be linked in any way. Identifying you as the person who provided any specific information for the study will not be possible. Because you will complete the survey anonymously, your name or other identifying information will not be used in reports or publications. Only the researcher may access study records to protect participants' safety and welfare.

If you have any questions, comments, or concerns regarding this study, contact me at [redacted] or by email at [redacted]. You can also contact the study's Dissertation Chairperson, Dr. Phil Pendley, by email at pendley@umassglobal.edu. If you have any further questions or concerns about your rights as a research subject, please contact UMass Global's Office of Institutional Research, UMass Global, 16355 Laguna Canyon Road, Irvine, CA 92618. BUIRB@umassglobal.edu.

Respectfully,

Christalle A. Hart Doctoral Candidate, UMass Global

APPENDIX C

Demographic Survey

RESEARCH STUDY TITLE: Designing and Delivering K-12 Education Professional Development: A Delphi Study

Lead Researcher: Christalle A. Hart, Doctoral Candidate UMass Global Department of Education [redacted], chart3@mail.umassglobal.edu

Faculty Sponsor:
Dr. Philip Pendley
UMass Global
Department of Education
951-712-2065, pendley@umassglobal.edu

- You are invited to participate in a research study to identify the delivery and content of K-12 educational professional development and the likelihood of the predictions being enacted by 2026. The main investigator for this study is Christalle A. Hart, Doctoral Candidate at the University of Massachusetts, Global (UMass Global) Doctor of Education in Organizational Leadership program. You were selected to participate in this study because of your expertise and experience in K-12 education and professional development.
- The purpose of this futures Delphi study was to determine what experts in professional development predict the delivery and content of K-12 education will look like in 2026 and beyond, to determine how the experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted professional development in 2026 and beyond.
- If you decide to participate in this study, you will receive three rounds of electronic surveys via Survey Monkey, with each survey taking approximately 15 to 20 minutes to complete. The Round 1 survey will contain an open-ended question. The Round 2 survey will utilize a Likert scale survey where participants will rate the recommendations that were identified from the first-round survey. Finally, the Round 3 survey will contain open-ended questions about the recommendations that were rated highest.

- This study involves no more than minimal risk, and no known harms or discomforts are associated with this study. There is no cost to you for participating, and you will not be compensated for your participation. The survey will be completed anonymously, and the researcher will not know your identity.
- Your participation in this study does not yield any direct benefits to you.
 However, analysis of the data generated from your participation in this study
 may contribute to the design and delivery of professional development in K-12
 education. Therefore, the information from this study is intended to inform
 researchers and educational leaders. Additionally, the findings from this study
 will be made available to all participants.
- Records of information that you provide for the research study, and any
 personal information you provide, will not be linked in any way. Identifying
 you as the person who provided any specific information for the study will
 not be possible. Because you will complete the survey anonymously, your
 name or other identifying information will not be used in reports or
 publications. Only the researcher may access study records to protect
 participants' safety and welfare.
- If you have any questions, comments, or concerns regarding this study, contact me at [redacted] or by email at [redacted]. You can also contact the study's Dissertation Chairperson, Dr. Phil Pendley, by email at pendley@umassglobal.edu. If you have any further questions or concerns about your rights as a research subject, please contact UMass Global's Office of Institutional Research, UMass Global, 16355 Laguna Canyon Road, Irvine, CA 92618. BUIRB@umassglobal.edu.

Do you agree to participate in this study? () Yes () No

Experience / Demographic Information

	for leading teams that design and deliver professional development? a. () Yes () No
2.	Please indicate your years of experience as a K-12 district level administrator. a. () Less than one year b. () At least one year, but less than two years c. () At least two years, but less than five years d. () Five years or more

1. Are you currently a K-12 district level administrator responsible, solely or in part,

3.	Please indicate the size of your school district.
	a. () 5,000 to 9,999 students
	b. () 10,000 to 19,999 students
	c. () 20,000 to 29,999 students
	d. () More than 30,000 students
4.	Please indicate the County in California where your school district is located.
	a. () Los Angeles County
	b. () Orange County
	c. () Riverside County
	d. () San Bernardino County
	e. () San Diego County

APPENDIX D

Informed Consent and Confidentiality

RESEARCH STUDY TITLE: Designing and Delivering K-12 Education Professional Development: A Delphi Study

UMASS GLOBAL 16355 LAGUNA CANYON ROAD IRVINE, CA 92618

RESPONSIBLE INVESTIGATOR: Christalle A. Hart, Doctoral Candidate

TITLE OF CONSENT FORM: Consent to Participate in Research

PURPOSE OF THE STUDY: This study is being conducted for a dissertation for the Doctor of Education in Organizational Leadership program at the University of Massachusetts Global (UMass Global). The purpose of this futures Delphi study was to determine what experts in professional development predict the delivery and content of K-12 Education will look like in 2026 and beyond, to determine how the experts in professional development rate the likelihood of the predictions made in Research Question 1 being enacted by 2026 and beyond, and identify how the experts predict K-12 educational organizations will need to structure themselves to deliver predicted professional development in 2026 and beyond.

PROCEDURES: In participating in this research study, I agree to either partake in three rounds of electronic surveys via Survey Monkey. The first-round survey will contain open-ended questions. The second round will utilize a Likert scale survey where participants will rate the recommendations identified from the first-round survey. Round 3 will contain open-ended questions about each of the recommendations rated highest.

I understand that:

- A. No known major risks or discomforts are associated with this research.
- B. Your participation in this study does not yield any direct benefits to you. However, analysis of the data generated from your participation in this study may contribute to the design and delivery of professional development in K-12 education. Therefore, the information from this study is intended to inform researchers and educational leaders. Additionally, the findings from this study will be made available to all participants.

Christalle A. Hart, UMass Global Doctoral Candidate, will answer any questions concerning my participation in this study. I understand that Mrs. Hart

may be contacted by phone at [redacted] or by email at [redacted]. The dissertation chairperson may also answer questions: Dr. Phil Pendley at pendley@umassglobal.edu.

- C. I understand that I may refuse to participate or withdraw from this study without any negative consequences. I also understand that the investigator may stop the study at any time.
- D. The study will utilize electronic surveys. All surveys and research data collected will be stored securely and confidentially on a password-protected server.
- E. No information that identifies me will be released without my separate consent, and all identifiable information will be protected to the limits allowed by law. If the study design or the use of the data is to be changed, I will be informed, and my consent re- obtained. If I have any questions, comments, or concerns about the study or the informed consent process, I may contact: UMass Global's Office of Institutional Research, UMass Global, 16355 Laguna Canyon Road, Irvine, CA 92618. BUIRB@umassglobal.edu.

I acknowledge that I have received a copy of this form and the Research Participant's Bill of Rights.

I have read the above and understand it and hereby v procedure(s) set forth.	roluntarily consent to the
Signature of Participant or Responsible Party	Date
Signature of Principal Investigator	Date

APPENDIX E

Participant Bill of Rights



UMASS GLOBAL INSTITUTIONAL REVIEW BOARD

Research Participant's Bill of Rights

Any person who is requested to consent to participate as a subject in an experiment, or who is requested to consent on behalf of another, has the following rights:

- 1. To be told what the study is attempting to discover.
- 2. To be told what will happen in the study and whether any of the procedures, drugs or devices are different from what would be used in standard practice.
- 3. To be told about the risks, side effects or discomforts of the things that may happen to him/her.
- 4. To be told if he/she can expect any benefit from participating and, if so, what the benefits might be.
- 5. To be told what other choices he/she has and how they may be better or worse than being in the study.
- 6. To be allowed to ask any questions concerning the study both before agreeing to be involved and during the course of the study.
- 7. To be told what sort of medical treatment is available if any complications arise.
- 8. To refuse to participate at all before or after the study is started without any adverse effects.
- 9. To receive a copy of the signed and dated consent form.
- 10. To be free of pressures when considering whether he/she wishes to agree to be in the study.

If you have questions regarding a research study at any time at any time, you should ask the researchers to answer them. You also may contact the UMass Global Institutional Review Board, which is concerned with the protection of volunteers in research projects. The UMass Global Institutional Review Board may be contacted either by telephoning the Office of Academic Affairs at (949) 341-9937 or by writing to the Vice Chancellor of Academic Affairs, UMASS GLOBAL, 16355 Laguna Canyon Road, Irvine, CA, 92618.

APPENDIX F

Survey Instrument Round 1

RESEARCH STUDY TITLE: Designing and Delivering K-12 Education Professional Development: A Delphi Study

What do experts in professional development predict the delivery and content of K-12 Education professional development will look like in 2026 and beyond?

APPENDIX G

Survey Instrument Round 2

RESEARCH STUDY TITLE: Designing and Delivering K-12 Education Professional Development: A Delphi Study.

The following is an example survey. The actual questions will be based on the responses provided by the participants in Round 1. Round 2 will include one question with a Likert rating and one open-ended question.

Based upon the first survey, the qualifications identified below as elements for delivery and content of K-12 education professional development.

Please rate the likelihood of the predictions made in Research Question 1 being enacted as it pertains to delivery and content of K-12 education professional development.

1. How do experts in professional development rate the likelihood of the predictions made in RQ1 being enacted by 2026 and beyond? \heartsuit o

	Very Likely	Moderately Likely	Likely Important	Lowly Likely	Not at all Likely
Prediction 1	\circ	\bigcirc	\bigcirc	\circ	\bigcirc
Prediction 2	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Prediction 3	\circ	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Prediction 4	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Prediction 5	\circ	\bigcirc	\bigcirc	\circ	\bigcirc
Prediction 6	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Prediction 7	\circ	\bigcirc	\bigcirc	\circ	\circ
Prediction 8	\circ	\bigcirc	\bigcirc	\bigcirc	\circ
Prediction 9	\circ	\bigcirc	\bigcirc	\bigcirc	\circ
Prediction 10	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\circ
Prediction 11	\circ	\bigcirc	\bigcirc	\circ	\circ
Prediction 12	\circ	\bigcirc	\bigcirc	\circ	\circ
Prediction 13	\bigcirc	\circ	\circ	\circ	\circ
Prediction 14	\circ	\circ	\circ	\circ	\circ
Prediction 15	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\circ

How do experts in professional development predict K-12 school districts will need to structure their organizations to implement the top five rated predictions for delivery and content of professional development?

APPENDIX H

Survey Instrument Round 3

RESEARCH STUDY TITLE: Designing and Delivering K-12 Education Professional Development: A Delphi Study.

The survey instrument used in Round 3 will contain an open-ended question for each of the highest-rated structures of professional development delivery and content revealed during Round 2.

The question will be: How do experts in professional development respond to the predictions made from RQ3 and what final suggestions do they make for implementation in 2026 and beyond?

APPENDIX I

UMass Global IRB Approval

IRB Application Approved As Submitted: Christalle Hart Indox x

 \square

ф

t

☆

Fri, Sep 9, 7:48 AM

Institutional Review Board <my@umassglobal.edu>

to me, pendley, irb 💌

Dear Christalle Hart,

Congratulations, your IRB application to conduct research has been approved by the UMass Global Institutional Review Board. This approval grants permission for you to proceed with data collection for your research. Please keep this email for your records, as it will need to be included in your research appendix. If any issues should arise that are pertinent to your IRB approval, please contact the IRB immediately at IRB@unnassglobal.edu. If you need to modify your IRB application for any reason, please fill out the "Application Modification Form" before proceeding with your research. The Modification form can be found at the following link: https://irb.umassglobal.edu/Applications/Modification.pdf.

Best wishes for a successful completion of your study.

Thank you,

Doug DeVore, Ed.D.

Professor

Organizational Leadership

ddevore@umassglobal.edu

www.umassglobal.edu

APPENDIX J

Collaborative Institutional Training Initiative-Certificate of Completion

